

MODELING EYE MOVEMENTS DURING DECISION MAKING: A REVIEW

MICHEL WEDELD

UNIVERSITY OF MARYLAND

RIK PIETERS

TILBURG UNIVERSITY

UNIVERSIDADE CATÓLICA PORTUGUESA

RALF VAN DER LANS

HONG KONG UNIVERSITY OF SCIENCE AND TECHNOLOGY

This article reviews recent advances in the psychometric and econometric modeling of eye-movements during decision making. Eye movements offer a unique window on unobserved perceptual, cognitive, and evaluative processes of people who are engaged in decision making tasks. They provide new insights into these processes, which are not easily available otherwise, allow for explanations of fundamental search and choice phenomena, and enable predictions of future decisions. We propose a theoretical framework of the search and choice tasks that people commonly engage in and of the underlying cognitive processes involved in those tasks. We discuss how these processes drive specific eye-movement patterns. Our framework emphasizes the central role of task and strategy switching for complex goal attainment. We place the extant literature within that framework, highlight recent advances in modeling eye-movement behaviors during search and choice, discuss limitations, challenges, and open problems. An agenda for further psychometric modeling of eye movements during decision making concludes the review.

Key words: eye movements, task switching, choice, search, decision making, hidden Markov model.

People make hundreds of decisions each day. For instance, car drivers search for traffic signs and decide to slow down or yield right-of-way (Ho, Scialfa, Caird, & Graw, 2001), airport security personnel search for weapons in X-ray images of luggage and decide whether to do a physical examination (McCarley, Kramer, Wickens, Vidoni, & Boot, 2004), radiologists search for nodules on mammograms and chest radiographs and decide which ones are potentially cancerous (Krupinski, Berger, Dallas, & Roehrig, 2003; Wedel, Yan, Siegel, & Li, 2016), people look at facial characteristics to decide which person they like (Chuk, Chan, Shimojo, & Hsiao, 2020), consumers search for products on websites and shelves (Shi, Wedel, & Pieters, 2013; van der Lans, Pieters, & Wedel, 2008b), search for information about prices, ingredients, and sustainability on packages (van Herpen & van Trijp, 2011), and choose the products they expect to satisfy their needs (Stüttgen, Boatwright, & Monroe, 2012). All these decisions involve search as well as choice. The underlying cognitive mechanisms rely heavily on acquisition and processing of visual information.

There is a long history of research into search and choice behaviors in psychology and related disciplines (examples are Berlyne, 1971; Russo & Rosen, 1975; Wolfe, 1998; Yarbus, 1967). This research has yielded a deep understanding of the underlying fundamental processes. Yet, three challenges face this literature. First, although many forms of decision making involve visual search, search and choice have been addressed in disparate streams of research, which

Correspondence should be made to Michel Wedel, Robert H. Smith School of Business, University of Maryland, College Park, MD 20742-1815, USA. Email: mwedel@umd.edu

© 2022 The Author(s)

has resulted in suboptimal cross-fertilization of knowledge. Second, much of the research in question has relied on endpoint measures, such as response times and the final choices made, to infer the underlying cognitive mechanisms of interest (Luce, 1977; McFadden, 1974; Treisman & Gelade, 1980; Wolfe, 1998). Such endpoint measures have proven to be valuable, but do not always provide unambiguous evidence for the underlying processes (Sanders & Donk, 1996; Zelinsky, 2008). Third, extant research has often relied on simple stimuli and controlled tasks in high-repetition, within-participant designs to uniquely identify the processes of interest. These controlled experiments ensure a high internal validity of hypothesis tests, yet may suffer from limited ecological validity of inferences on the cognitive processes as they play out in real-life contexts.

There is growing interest in understanding search and choice in the common, more complex, messy, and less controlled tasks that people encounter in their daily lives. Such research faces critical challenges because search and choice behaviors are often involved in the same task, and multiple unobserved cognitive processes may simultaneously cause variations in the associated endpoint measures. Eye movements are unique process measures to study search and choice behavior in such natural contexts (Findlay & Gilchrist, 2003; Najemnik & Geisler, 2005; Zelinsky, 2008). Eye movements reflect, with a high temporal and spatial resolution, several unobserved perceptual, cognitive and evaluative processes (Findlay & Gilchrist, 1998). They enable a fine-grained process analysis with the potential of yielding insights that are difficult to obtain otherwise, especially in real-life contexts. For example, eye movements have provided insights into information search during advertising exposure (Wedel & Pieters, 2000) and brand choice (Shi et al., 2013), the determinant processes of which are largely inaccessible to self-reports (Aribarg, Pieters, & Wedel, 2010; Nisbett & Wilson, 1977) or to alternative process-tracing techniques such as information display boards (Lohse & Johnson, 1996). Statistical models have proven to be indispensable for making inferences on these unobservable processes, in psychology, economics and marketing, because they enable disentangling the effects of multiple unobserved processes from the eye-movement recordings (Stüttgen et al., 2012; van der Lans et al., 2008b; Yang, Toubia, & de Jong, 2015). Understanding the mechanisms that drive search and choice may lead to better predictions and has ramifications for policy makers, companies, and consumers, for example by enabling better design and testing of policy and medical interventions, websites, store shelves, advertisements, and labels on food packaging (e.g., Van Loo, Grebitus, Nayga Jr., Verbeke, & Roosen, 2018).

Several streams of research have examined search and choice behaviors in natural environments by using eye-movement measures and statistical, psychometric, or econometric models. Earlier reviews have summarized advances in eye-tracking research, in general (Rayner, 1998; Wedel & Pieters, 2008a) and for search and choice separately (Glaholt & Reingold, 2011; Kowler, 2011; Orquin & Loose, 2013), or have conducted meta-analyses of eye movements during choice (Orquin, Lahm, & Stojić, 2021). The present article reviews the advances made in the modeling of eye movements during decision making. We aim to make a step toward an integrated account of the cognitive processes fundamental to decision making by highlighting the role of task and strategy switching during such decision making. We provide a theoretical framework that summarizes the key tasks involved and the processes underlying eye movements during decision making. We place the extant literature within that framework, highlight recent advances in modeling eye-movement behaviors, discuss limitations, challenges, and open problems, and provide an agenda for future research. We believe that such a review of the literature is timely, because the cost of eye-tracking equipment has come down enormously and relatively unobtrusive eye-movement recording is now possible in a wide variety of real-life contexts, while the psychometric toolbox to extract information from eye-tracking data is rapidly growing. Even so, eye-movement recording and modeling during search and choice is still underutilized both in academic and in applied research, a situation which we hope this review will help to improve.

Section 1 briefly introduces eye movements and eye-movement recording. Section 2 presents our framework of tasks, task switching, and attentional processes underlying eye movements during decision making. Section 3 summarizes models for each of those processes during search. Section 4 reviews, against this backdrop, the literature on eye-movement analysis during choice. Section 5 provides an outlook for novel applications and future developments.

1. Eye Movements and Their Recording

1.1. Eye Movements and Visual Attention

We use the term (visual) attention as a container for various cognitive processes that are involved in the selection of relevant and the suppression of irrelevant information during search and choice, for which according to James (1890, pp. 403–404): "Focalization, concentration, and consciousness are of [...] essence." The next section describes these processes and their links to eye movements. It is important to point out that in most natural decision-making tasks, eye-movement patterns are valid and accurate indicators of the hidden attention processes of interest (Findlay, 2005). There are also successful examples of inferences on higher order cognitive states from eye movements, such as on tasks and goals (Borji & Itti, 2014; Haji-Abolhassani & Clark, 2014; Kardan, Berman, Yourganov, Schmidt, & Henderson, 2015) and on the emotional valence of images (Nummenmaa, Hyönä, & Calvo, 2006; R.-Tavakoli et al., 2015), although these tend to require additional experimental control, measures, and theoretical assumptions.

At any point in time, the human eye-vision system processes only about one percent of the field of vision with high acuity (Rayner, 1998; Wedel & Pieters, 2008a). This is due to the density of photosensitive cells in the human retina decreasing rapidly from a high concentration in a small region in its center (fovea) toward the parafovea and periphery. Therefore, people need to move their eyes to explore their environment in sufficient detail. In practice, people make on average three to six larger eye movements per second during tasks such as reading, scene perception, or decision making (Rayner, 1998). These saccades are fast jumps of the eye's point of regard (POR) between spatial locations in the environment. Saccades typically last for about 20–40 ms. Fixations are periods of about 100–400 ms when the eyes hardly move. While during a saccade visual perception is suppressed to prevent blurring, during an eye fixation detailed visual information is obtained from a small region of about two degrees of visual angle around the POR.

A scan-path is the spatiotemporal sequence of eye fixations and saccades for a particular task, stimulus, and person. Figure 1 summarizes the scan-paths of four participants who were engaged in a target search task: they searched on a shopping website for a brand of perfume to choose as a gift. The figure (based on data from study 2 in van der Lans, Pieters, & Wedel, 2021) reveals several key features of the scan-paths which inform theories and psychometric models. First, the first fixation commonly lands close to the center of the screen. This location may be optimal because it has the smallest distance to all other locations. Second, some fixations land on visually salient objects, i.e., objects that stand out from their surroundings in terms of luminance, color, or shape. Other parts of individuals' scan-paths reflect systematic left-right movements. That pattern may be induced by the organization of the image (scene layout). Third, the scan-paths consist of clusters of fixations on objects, interspersed with saccades toward more distant objects. Those clusters reflect refixations which are used to accumulate information to determine the identity or value of objects of interest. Longer saccades reflect search for where new objects are located. Fourth, a cluster of fixations is specifically directed at the final, chosen object. Such a pattern may reflect preference formation or a final verification of the object before implementing a decision, or it might be that looking at an object longer increases one's preference for it. Fifth, scan-paths show large individual variation and differ markedly among distinct tasks and stimuli. In addition,

the scan-paths of participants who had a different search goal (in the top versus the bottom row of the figure) are noticeably different. We return to all those features of scan-paths in the sequel. The scan-paths observed when people make choices among alternative options are similar to those in Fig. 1 but are often considerably more elaborate.

Next to fixations and saccades, other types of eye movements include micro-saccades which are involuntary movements with small amplitude that occur during fixations, smooth pursuits which are fixations on moving objects, vergence movements which occur to maintain the POR on an object that moves toward or away from the viewer, and pupil movements to regulate the amount of incoming light, amongst others. While the analysis of these movements has been useful for specific problems, the present review focuses on fixations and saccades.

1.2. Eye-Movement Recording

The dominant method of recording eye movements in research on search and choice uses an infrared light source directed at one or both eyes, which creates (invisible) reflections on the cornea, the hard outer layer of each eye, called Purkinje reflections (Duchowski, 2003). After a short calibration task, video cameras record one or more of these reflections, even while participants move their heads. Based on the calibration results, algorithms infer the POR in x-y stimulus coordinates from these reflections. Commonly used eye trackers are built into desktop monitors or into small stand-alone devices that can be attached to laptops and other objects. In addition, mobile eye trackers embedded in glasses allow free body movements while recording the entire field of view as well as the POR within it. Commercial infrared eye trackers typically sample the POR with a sampling frequency of 50 or 60 Hertz and a spatial accuracy of 0.5 degrees of visual angle, or better. Eye trackers with a higher spatiotemporal precision make use of multiple Purkinje reflections and sampling frequencies of 120 Hertz or more, for which the participant's head sometimes needs to be fixed during recording with a chin or forehead rest. Further, eye movements can be recorded with regular front-facing (web) cameras integrated into digital devices, by using eye landmarks, such as the center and shape of the pupil and the location of eye corners, as input to computer vision algorithms that estimate gaze direction from images of the eyes. Those eye-tracking solutions still have lower accuracy than infrared eye tracking.

1.3. Eye-Movement Data

Eye tracking during decision making results in large amounts of raw data. For example, an eye-tracking study in which participants make 20 decisions, with 10 s available for each decision and a 50 Hertz sampling frequency, results in some ten thousand samples comprised of the x-y coordinates of the POR of each eye per participant. This amounts to one million data points for a study with one hundred participants. A proportion of the data may involve outliers or may be missing due to eye blinks and tracking problems, however. From the raw samples of the POR, algorithms are used to identify fixations and saccades based on their spatial proximity or the velocity of their movement (Salvucci & Goldberg, 2000). This reduces the dimensionality of the data by a factor 20 or more. For example, the BIT (Binocular Individual Threshold) algorithm (van der Lans, Wedel, & Pieters, 2011) automatically determines fixations using individual-specific eye-velocity thresholds for both eyes, automatically removing eye blinks and other recording abnormalities. Due to blinks, saccades and recording errors, missing data points are common in raw samples of the POR. As a rule of thumb, van der Lans and Wedel (2017) propose that one may consider removing a participant's data if less than 80 percent of their raw samples are classified as fixations. Once the fixation data have been prepared, various characteristics of the eye-movement patterns can be examined, such as x-y coordinates of fixations over time, scan-paths, or aggregate measures such as fixation counts, gaze or dwell times, and fixation selection indicators, on socalled Regions of Interest (ROIs) (Holmqvist et al., 2011). The spatial aggregation of fixations



FIGURE 1.

Scan-paths of four participants searching for a brand of perfume to choose on a shopping website. The dots indicate fixations, the lines saccades. Red rectangular boxes indicate the first and last fixation. The top, respectively, bottom, panels show scan-paths for two participants searching for the Jimmy Choo, respectively, the Dolce & Gabana, perfumes (based on data from van der Lans et al., 2021).

on those ROIs, for example by calculating gaze times on the individual perfume bottles in Fig. 1, reduces the size of the data further (the gaze time on a ROI is defined as the sum of the durations of all fixations that land on that ROI). Fixation patterns are often graphically explored via heatmaps, which represent the density of fixations using colors, and fixation plots which depict scan-paths as shown in Fig. 1.

2. Theoretical Framework

Via eye movements, people acquire information to reduce various uncertainties during task completion (Hayhoe & Ballard, 2005; Land, 2019; Pieters & Wedel, 2007). Decision-making tasks vary in the types of uncertainty that they induce, which influences the eye-movement patterns observed during these tasks. We propose a classification of decision-making tasks in Sect. 2.1 and, building on this, a theoretical framework of eye movements during decision making in Sect. 2.2.

2.1. Decision-Making Tasks

In Table 1, we propose a classification that involves six fundamental decision-making tasks that people commonly engage in. These tasks have in common that they require eye movements. They therefore involve exposure durations longer than the duration of a single fixation (100–300 ms). The tasks elicit different types of uncertainties which people try to resolve through visual information acquisition.

Table 1 makes a main distinction between *perceptual decision making* and *preferential decision making* (e.g., Dutilh & Rieskamp, 2016; Summerfield & Blangero, 2017).

Downloaded from https://www.cambridge.org/core. 07 Jan 2025 at 18:39:35, subject to the Cambridge Core terms of use.

I. Ta	usk	II. Uncertainty	III. Example	IV. Dominant Process
Perc M	ceptual Decision aking: Search			
1	Localization	Space	Where in the scene is object j, if at all?	Perception
2	Identification	Object	Is this object j or not, or is it object k?	Perception
3	Specification	Attribute	Does object j have attribute m, and how much of it?	Cognition
Pref M	erential Decision aking: Choice			
4	Inference	Outcome	Will attribute m of object j lead to outcome y for person i?	Cognition
5	Evaluation	Value	How valuable or meaningful is attribute m (of object j) to person i?	Evaluation
6	Integration	Utility	How much does person i value object j (over object k)?	Evaluation

TABLE 1. Six fundamental decision-making tasks, the types of uncertainty they induce, and the dominant cognitive process involved when engaging in the task.

Perceptual decision-making tasks have an objective (external) performance criterion, mostly speed and/or accuracy. Search tasks are perceptual decision-making tasks. Preferential (or value-based) decision-making tasks have one or more subjective (internal) performance criteria, such as perceived costs and benefits, decision justification, or post-choice satisfaction. The completion of these tasks is based on personal preference or utility functions. Choice tasks are preferential decision-making tasks. The dominant processes involved in the tasks in Table 1 are perception, cognition, and evaluation, respectively (Table 1, column IV). Although cognition is sometimes used in a broader sense to encompass perception and evaluation, here *perception* refers to the process of acquiring new information via the senses, *cognition* refers to the processes of storing, retrieving, interpreting, and combining new and existing information, and *evaluation* refers to assessing the personal value of the information (and acting upon it).

We distinguish six basic tasks, each having its own specific type of uncertainty that people need to reduce (Table 1, columns I and II), and each being at least partly observable from eye movements. 1. *Localization* aims to reduce uncertainty about the spatial location of a known object; 2. *Identification* aims to reduce uncertainty about the identity of an object vis-à-vis other objects and/or some object template in memory; 3. *Specification* aims to reduce uncertainty about the presence of specific attributes or features of an object by associating the object with information on these attributes in memory; 4. *Inference* aims to reduce uncertainty about the consequences of these attributes of an object's attributes and features to the person's goals; 6. *Integration* aims to reduce uncertainty about the overall utility of an object (choice alternative) to the person relative to other objects. We use the term "integration," because people need to integrate expected outcomes and their valuation of these outcomes into their overall assessment of the utility of a stimulus (Anderson & Shanteau, 1970). Integration can take place via some (optimal) subjective expected utility mechanism, as traditionally assumed in economics, or via satisficing mechanisms

(Simon, 1955). The order of uncertainties in the table from top to bottom corresponds to the hierarchical organization of knowledge structures from bottom-up features of objects to topdown values and preferences of people. It reflects the structure of the human motivation system from concrete means (objects and their attributes), which have expected consequences (outcomes), which satisfy certain abstract ends (values) (Kruglanski, Chernikova, Babush, Dugas, & Schumpe, 2015), which contribute to utility.

The tasks that people engage in in real-life contexts are commonly combinations of these six basic tasks. For example, localization (task 1) and identification (task 2) are subtasks of a target search task (van der Lans et al., 2008b). Localization (task 1) and specification (task 3) are subtasks of an information search task (Moorthy, Ratchford, & Talukdar, 1997). Identification (task 2) and specification (task 3) are subtasks of a categorization task (Rosch, 1978), in which people differentiate prototypes from exemplars in a category or exemplars in one category from exemplars in another. It is crucial to note that whereas controlled experiments can be used to isolate a single task or uncertainty, the tasks that people perform in natural settings are mostly comprised of several subtasks, each with their specific uncertainty (Droll & Hayhoe, 2007; Hayhoe & Ballard, 2005; Locke & Latham, 2002).

2.2. Theoretical Framework

The classification in Table 1 informs our framework, summarized in Fig. 2, of how people use eye movements to reduce uncertainties inherent in perceptual and preferential decision-making tasks. The framework relates eye movements to the underlying processes of interest. It shows from left to right 1) inputs, 2) processes, and 3) decision outputs. These concepts are organized from bottom to top by level of processing, from features, to scene, objects, and the person. Symbols indicate the links between these concepts.

First, the framework distinguishes five observable exogenous inputs: a) person characteristics (i.e., age, prior knowledge), b) task instructions (i.e., search goal, incentives, time constraints), c) object attributes (i.e., description of objects/products on dimensions such as price, brand name and other qualities), d) scene layout (i.e., how the objects are arranged on a row-column shopping website), and e) perceptual features (i.e., detailed visual characteristics, such as colors and edges of objects). Second, the outputs of decision tasks consist of observable micro- (fixation locations, fixation durations) and macro-decision outcomes (search accuracy, choice, and response time), which are represented at the right-hand side of Fig. 2. Third, in between these observable inputs and outputs are unobserved, or latent, processes, depicted in the center of the figure (connected via P1-P5). These latent processes are affected by exogenous inputs (via links I1-I5), and in turn impact both eye movement (via O1-O2) and choice decisions (via O3-O4).

Task and strategy switching take a central role in the framework. Formally, task switching concerns switching between two or more tasks which each have their own unique goals and uncertainties (Kiesel et al., 2010), whereas strategy switching concerns switching between different ways of pursuing the same task or goal (Shi et al., 2013). In this review, we use the terms interchangeably, because in real-life contexts, empirically the distinction between task and strategy switching is often ambiguous. Decision making in real-life contexts typically involves the deployment of attention to resolve multiple uncertainties over time (Table 1). Therefore, people break up complex decision tasks into multiple simpler subtasks, each of which requires a unique strategy that involves the deployment of attention to resolve the deployment of attention, & Matthews, 2016; Stojić, Orquin, Dayan, Dolan, & Speekenbrink, 2020). People switch between these subtasks over time to assess the utility of decision alternatives and meet the overall decision goal, under cognitive cost and effort constraints. They monitor goal progress within tasks, and switch when a task nears completion and/or incremental expected utility drops below or cost rises above a threshold (Gutzwiller, Wickens, & Clegg, 2019). Such





Eye movements and decision task performance. Framework for how people use eye movements to reduce uncertainties inherent in perceptual and preferential decision-making tasks. Observable exogenous inputs to the decision processes are on the left side; unobservable perceptual, cognitive, and evaluative processes are in the center; observable decision outcomes are on the right-side. Inputs and processes are organized from top to bottom by level of visual processing: person, object, scene, and feature. Arrows depict the links between inputs and processes (I1 to I5), between unobserved processes (P1 to P5), and between processes and decision outputs (O1 to O4).

task/strategy switching characterizes goal pursuit and decision making in complex, realistic contexts (Locke & Latham, 2002). The next sections describe the attention processes during decision making. Appendix 1 summarizes the neuroscientific basis of attention and eye movements.

2.2.1. Eye Movements and Attention Overt (observable) eye movements are tightly coupled with covert (unobservable) shifts of visual attention (Fig. 2, O1). Attention and eye movements are guided by the same regions in the brain (Corbetta & Shulman, 2002; Findlay, 2005; Findlay & Gilchrist, 2003; Appendix 1), and their coupling has been likened to a rubber band (Henderson & Hollingworth, 1998): "the eyes go where attention goes and attention goes where the eyes go". Therefore, and because they are subject to (mechanical) measurement errors (Duchowski, 2003), eye-movement recordings are probabilistic indicators of the location and duration of covert attention (Baddeley & Tatler, 2006).

2.2.2. Task and Strategy Switching Most real-life decision tasks require the reduction in more than one type of uncertainty (Table 1). Rather than reducing these uncertainties simultaneously, during real-life decision making people switch between attention/decision strategies that each aim at reducing a specific type of uncertainty (Haji-Abolhassani & Clark, 2013, 2014; Liechty, Pieters, & Wedel, 2003; van der Lans, Pieters, & Wedel, 2008a; van der Lans et al., 2008b; Wedel, Pieters, & Liechty, 2008). Because the human visual system uses eye movements to acquire information "just in time" when the demands of the current task call for it (Hayhoe & Ballard, 2005), shifts in these strategies are reflected in the observed scan-path of eye movements (Mayr, Kuhns, & Rieter, 2013).

Strategy shifts may occur because of competition between subtasks. Task schemas that drive attentional processes may exert mutually inhibitory effects (Gilbert & Shallice, 2002). Top-down (executive) control may also be involved to selectively activate or deactivate these schemas,

depending on which subtask has gained priority (Logan & Gordon, 2001). Such subconscious task interference and volitional top-down control may both operate at the same time (Kiesel et al., 2010). Strategy switching is thus central to decision making in complex real-life settings (Venkatraman, Payne, & Huettel, 2014). We submit that it is necessary for implementing both micro- (eye movement; Fig. 2, O1) and macro-(search and choice; Fig. 2, O3) decisions that facilitate the attainment of the overall task goal. Instances of attentional strategy switching are (1) processing the location versus the identity of objects during target search (van der Lans et al., 2008a; 2008b), (2) processing by object across attributes versus by attribute across objects during information search and choice (Shi et al., 2013), (3) systematic versus salience-based attention allocation during target search (van der Lans et al., 2008a; 2008b), and (4) habitual versus goal-directed processing during choice (Ursu, Zhang, & Erdem, 2021). To illustrate, we expand on the first two of these processes, which occur during perceptual (location/identity) and preferential (by attribute/object) decision making, respectively. Sections 3.1 and 3.2 describe these processes in more detail.

First, during perceptual decision making, visual information processing in the human brain takes place in two broad pathways (Glaholt & Reingold, 2011; Ungerleider & Mishkin, 1982; Appendix 1), each tailored to reduce a specific type of uncertainty (Table 1). The "what" pathway is involved in identification of objects (faces, hands, people, houses). The "where" pathway is involved in the localization of objects. The "what" (identification) and "where" (localization) pathways give rise to different oculomotor patterns (Bullier, Schall, & Morel, 1996; Pannasch & Velichkovsky, 2009; Appendix 1). Activity in the "what" stream tends to produce short saccades (Liechty et al., 2003; Pannasch & Velichkovsky, 2009), with fixations that cluster in a small number of regions that are deemed informative to the task (Smith & Henderson, 2009; Wedel et al., 2008; Yarbus, 1967). Those short saccades result in repeated fixations on an object, which minimizes working memory load and may be required for target identification and specification of the object in terms of its attributes (Droll & Hayhoe, 2007; Rayner, Smith, Malcolm, & Henderson, 2009; Smith & Henderson, 2009). Tasks with more complex stimuli (naturalistic versus abstract stimuli) and a higher working memory load (choice versus search tasks) therefore typically result in more and longer fixations (Gould, 1973; Orquin & Loose, 2013). Activity of the "where" stream tends to produce longer saccades that serve to rapidly bring peripheral, salient locations and objects into focus. Eye-tracking research has revealed that people frequently switch back and forth between such "what" and "where" strategies, which, respectively, involve repeated fixations on a small region, and longer saccades between distant regions (Liechty et al., 2003; Pannasch & Velichkovsky, 2009; Wedel et al., 2008).

As a second example, during preferential decision making on attribute-by-product matrices, people acquire information using processing-by-attribute or processing-by-product strategies (Bettman, Luce, & Payne, 1998; Payne, Bettman, & Johnson, 1993). Attribute-based processing involves extraction of information on a single attribute across multiple products. Such processing involves a low-effort attribute-specification strategy that serves to compare products. Product-based processing involves acquisition of information on a single product across multiple attributes. It is a high-effort strategy which supports the integration of information into an overall expected value or utility (Martinovici, Pieters, & Erdem, 2021). Early process-tracing methods such as Mouselab, where participants open and close information cells on an electronic display, suggested that people first use by-attribute processing and then switch to by-product processing before making a choice (Bettman et al., 1998). Eye-tracking research initially confirmed that during preferential decisions people first tend to make more inter-product saccades which reflect processing-by-attribute, and then more intra-product saccades which reflect a processingby-product strategy (Pieters & Warlop, 1999). However, rather than reflecting a simple and orderly two-stage process, eye-movement data have demonstrated that people repeatedly switch between these two strategies (Table 1) even during decision-making tasks that last a few minutes at most. People tend to extract information on two or three products in a by-attribute strategy, and on two or three attributes in a by-product strategy, and switch back and forth between these two strategies multiple times before making a choice (Shi et al., 2013).

2.2.3. Bottom–Up Factors Bottom-up factors residing in the stimulus affect attention processes and thus eye movements (Fig. 2, P3), and exert a large influence especially during object localization. These visual factors have strong effects on attention, comparable in size to those of top-down factors such as task instructions (Orquin et al., 2021). Basic perceptual features, such as luminance, edges, contours, and colors (Treisman & Gelade, 1980; Wolfe, 1994), are extracted and combined into a salience map, or attention priority map (Fig. 2, II; Appendix 1) (Itti & Koch, 2001; Itti, Koch, & Niebur, 1998; Koch & Ullman, 1985). The attention priority map represents the conspicuousness of locations in the visual field and guides attention and eye movements (Fig. 2, P3) (Donk & Soesman, 2010; Treisman & Gelade, 1980), because the focus of attention successively shifts to locations on the map with decreasing priority (Foulsham & Underwood, 2008; Parkhurst, Law, & Niebur, 2002). Perceptual *pop-out* occurs when a particular location in the visual field stands out due to a basic feature that draws attention almost immediately (Parkhurst et al., 2002; Treisman & Gelade, 1980). Inhibition of return (IOR; Klein, 2000; Posner & Cohen, 1984) encourages exploration of novel locations by temporarily inhibiting the return of the eyes to previously attended locations or objects (Appendix 1).

The global layout of the scene also influences eye movements via what is called *contextual guidance* (Fig. 2, I2) (Torralba, Oliva, Castelhano, & Henderson, 2006). The essential meaning of a scene, or its gist, is extracted bottom up based on the spatial distribution of low-level features (Oliva & Schyns, 2000; Oliva & Torralba, 2006). This process is fast. The gist of a typical scene can be perceived even within a single fixation, in less than 100 msec. (Pieters & Wedel, 2012; Rousselet, Joubert, & Fabre-Thorpe, 2005), presumably immediately upon exposure to it. Gist perception helps the localization and identification of objects and guides the initial scan-path (Mack, Gauthier, Sadr, & Palmeri, 2008; Appendix 1).

During search tasks on visual scenes, fixating on the center of the scene first may be optimal for assessing the gist and basic features across the entire visual field (Itti & Koch, 2001; Koch & Ullman, 1985). This optimal viewing position facilitates subsequent eye movements to salient or informative locations (Tatler, 2007). Therefore, there is a tendency to begin the scan-path close to the center of the image or computer screen on which the scene, for example an advertisement or shopping website (Fig. 1), is presented (Mannan, Ruddock, & Wooding, 1995; Parkhurst & Niebur, 2003; Reinagel & Zador, 1999; Tatler, 2007). More generally, the layout of visual displays has been shown to affect information acquisition processes and the direction of the scan-path (Glaholt, Wu, & Reingold, 2010; Pieters & Warlop, 1999; Shi et al., 2013; Fig. 2, 12).

2.2.4. *Top–Down Factors* Top-down factors, residing in the person and the task, such as memory (Olivers, Meijer, & Theeuwes, 2006) and goals (Hayhoe & Ballard, 2005; Pieters & Wedel, 2020), affect eye movements in an interplay with bottom-up processes (Fig. 2, P4, P5). Yarbus (1967) first showed how eye movements are dramatically directed toward stimuli that are informative for the current task or goal. Attention to visual features the viewer believes to be instrumental for task performance is enhanced, and attention to features that are deemed irrelevant is suppressed, via top-down influences on the attention priority map (Pieters & Wedel, 2007; Fig. 2, P5; Appendix 1).

Top-down processes may also be primed by the global layout of a visual scene, which may lead to orderly spatial sequences of eye movements that reflect individuals' use of systematic search and processing strategies (Monk, 1984; Ponsoda, Scott, & Findlay, 1995; Spalek & Hammad, 2005). These occur, for example, when a natural scene with a horizontal layout of objects or text

primes the first fixation to be at the top-left, and the saccades to be predominantly from left to right (Shi et al., 2013).

Practice and time pressure are also key top-down factors (Fig. 2, I4). Practice improves decision performance by making the inhibition of distracting stimuli more efficient. Practice results in (1) improved speed of locating and discriminating targets from distractors, i.e., in faster reduction in location and identity uncertainty, and in (2) shorter fixation durations and thus faster response times (response times during decision tasks are approximately equal to the sum of all fixation durations; van der Lans et al., 2021; Zelinsky & Sheinberg, 1997). When making decisions under time pressure, people switch from more complete full-information search strategies to more efficient partial-information search strategies (Pieters & Warlop, 1999) which may involve (1) shortening fixation durations, or (2) filtering a subset of the available information by ignoring certain objects or their attributes, or (3) shifting to a different acquisition strategy all together (Fig. 2, P4). Specifically, people may switch from a processing-by-product, characterized by intraproduct saccades, to a processing-by-attribute strategy, characterized by inter-product saccades. Individual differences in eye movements may be due to a variety of unobservable states and traits and have been documented in reading, scene viewing, and various types of decision tasks (Fig. 2, 15; e.g., Henderson & Hollingworth, 1998; Lee & Webb, 2005; Pieters & Wedel, 2007; Rayner, 2009; Rutishauser & Koch, 2007; Shen & Palmeri, 2016).

2.2.5. Utility/Value Utility, or subjective value, is a measure of the happiness or satisfaction that consumers get from searching, acquiring and/or experiencing goods and services. Economic decisions are often assumed to be maximizing utility: the best choice outcome provides the highest expected utility to the decision maker (McFadden, 1974). Several dual-system accounts of decision making have been proposed for value learning and value-based decision making (Damasio, 1994; Daw & O'Doherty, 2014; Kahneman, 2011; Wang, 2002), whereby reflexive versus reflective decisions, compensatory versus satisficing decisions, or decisions pertaining to positive versus negative departures from a baseline, are controlled by mutually inhibiting (top-down) systems (Fig. 2, P4).

The empirical association between eye-movement measures of visual attention and value or choice outcomes is well established (Isham & Geng, 2013; Krajbich, Armel, & Rangel, 2010; Pieters & Warlop, 1999; Schotter, Berry, McKenzie, & Rayner, 2010; Shi et al., 2013; Shimojo, Simion, Shimojo, & Scheier, 2003; Stewart et al., 2016; Stojić et al., 2020). Evidence for the role of attention in choice comes from studies that use statistical mediation analysis to assess the extent to which the effect of visual factors on choice behavior is mediated via visual attention, by calculating their indirect effects through the mediator (Zhang, Wedel, & Pieters, 2009). Those studies have looked into the effects of visual marketing factors such as package design (Milosavljevic, Navalpakkam, Koch, & Rangel, 2012), nutrition information (Bialkova & van Trijp, 2011), shelf position (Atalay, Bodur, & Rasolofoarison, 2012; Chandon, Hutchinson, Bradlow, & Young, 2009; Chen, Burke, Hui, & Leykin, 2021; Deng, Kahn, Unnava, & Lee, 2016), assortment (Townsend & Kahn, 2013), advertisement displays (Zhang et al., 2009) and product salience on websites (van der Lans et al., 2021). This stream of research has provided consistent evidence that the effects of visual factors on search and choice outcomes are statistically mediated by eye-movement metrics.

One significant top-down effect occurs when a previously rewarded stimulus continues to capture attention automatically even if it is no longer associated with the reward (Fig. 2, P5) (Anderson, Laurent, & Yantis, 2011; Della Libera & Chelazzi, 2006). The learned stimulus-reward association results in an enduring attentional priority and a very slow extinction response (Anderson & Yantis, 2013) that hampers suppression of that stimulus during subsequent search and choice for task-relevant stimuli.

Importantly, the relationship between utility and attention is bidirectional (Fig. 1, P1 P2). Despite some alternative accounts (Glaholt & Reingold, 2011), there is converging evidence that people tend to look more at what they like (Fig. 2, P1; Callaway, Rangel, & Griffiths, 2021; Gluth, Kern, Kortmann, & Vitali, 2020) and tend to like more what they look at (Fig. 2, P2; Bhatnagar & Orquin, 2021; Gluth et al., 2020). Shimojo and co-authors (2003) first proposed this bidirectional effect as a mechanism for the *gaze cascade*, which is the rapidly accumulating attention on the chosen alternative just before (500–750 ms) a choice is expressed. The gaze cascade is a robust phenomenon that has been demonstrated to occur in two-alternative perceptual and preferential decision tasks (Glaholt & Reingold, 2009), multi-attribute choice (Atalay et al., 2012), gambles (Fiedler & Glöckner, 2012), and intertemporal decision making (Fig. 2, P1; Fisher, 2021).

3. Perceptual Decision Making: Eye-Movement Models of Search

In *target search*, people search for a predefined object surrounded by other objects on a visual display (Table 1). Search for a specific perfume on the website in Fig. 1 is an example. Target search may occur before people have made a choice, such as when searching for products to consider buying, or after they made a choice, such as when searching for a product they have earlier decided to buy. In *specification search*, people search for information on the attributes of an object (Table 1), such as the price, brand name or color, among information about other objects (Moorthy et al., 1997). We next discuss models of these two types of perceptual decisions (Appendix 2 has modeling details).

3.1. Eye-Movement Models of Target Search

In Hidden Markov Models (HMM), unobserved discrete states that evolve over time according to a Markov Process are assumed to generate the observed eye-movement time series according to some probabilistic model. Over-time variations in the observed eye movements are reflections of these unobserved states and the switches between them. Liechty et al. (2003) were among the first to model latent attention states via HMMs to identify shifts between attention strategies that people use during exploratory search upon advertising exposure.

During search for complex targets, people aim to reduce two types of uncertainties: location and identity uncertainties (Ungerleider & Mishkin, 1982; Appendix 1; Table 1). They do so by switching over time between two attention states in which one of the two subtasks takes precedence (see Liechty et al., 2003). The two hidden states, localization and identification, are assumed to follow a Markov process over time. Van der Lans et al. (2008a; 2008b) developed HMMs to uncover the time path of localization and identification states from eye-movement patterns during target search. The models in question specify the location of each fixation, in pixel coordinates, as a spatial point process. The two attention states are identified via parametric assumptions on that process that are specific to each state. The localization state causes longamplitude saccades and the identification state short-amplitude saccades (Bullier et al., 1996; Thompson, 2005; Appendix 1). Eye movements in the localization state are driven by the salience of locations in the visual display (Itti & Koch, 2001; Koch & Ullman, 1985; Appendix 1). In the model, salience is defined in terms of pixel-level perceptual features, including colors, brightness, and edges. Computer vision techniques are used to extract these features from digital images of the search display, resulting in (RGB or CIELAB) feature values for each pixel. Because the eye only processes detailed information from a region of about two degrees of visual angle around the point of fixation (Appendix 1), a Normal spatial kernel with a width of two degrees is used to smooth these pixel-level variables. The salience map is then represented as a weighted combination of perceptual features, the weights being represented by individual-level model parameters. These parameters are assumed to follow a Normal distribution and are estimated based on the eye-movement data. They capture the effect of top-down factors on attention (Fig. 2, P5). Although some consider salience to be a purely bottom-up stimulus property (Itti & Koch, 2001), its operationalization as a weighted sum of basic features by Van der Lans et al. (2008a; 2008b) is in line with the idea that salience results from activation or inhibition of visual information that is deemed, respectively, relevant or irrelevant, which is a top-down process (Gaspelin & Luck, 2018; Sawaki & Luck, 2010). Similar probabilistic formulations of salience had been previously proposed (Baddeley & Tatler, 2006; Rutishauser & Koch, 2007). The Van der Lans et al. target search model provides estimates of individual-level salience maps. Further, in the localization state, factors measured at the level of the display (scene organization or display architecture) and the person (task schemas or scanning habits) drive eye movements (Fig. 2, I2, 15). Systematic eye-movement strategies (Spalek & Hammad, 2005) are captured via Markov transition probabilities on the ROIs containing objects to the left and right, or top and bottom, respectively, of the previously fixated ROI. The identification state accommodates the tendency to re-fixate on an object to collect more information on its identity (Smith & Henderson, 2009; Tatler & Vincent, 2009). This model and its extensions (van der Lans et al., 2021) incorporates a range of well-documented neuro-psychological processes (Appendix 1).

Top-down modulation of low-level visual processes involves selectively enhancing visual features that are diagnostic for a search task and suppressing features that are non-diagnostic (Desimone & Duncan, 1995; Fig. 2, P5; Einhäuser, Rutishauser, & Koch, 2008; Rutishauser & Koch, 2007). Bayesian formulations that encapsulate top-down information via a hierarchical prior reflect the mechanism involved (Borji, Sihite, & Itti, 2014; Lee & Mumford, 2003; Torralba et al., 2006). Van der Lans et al. (2008a) manipulate task instructions to enable the decomposition of bottom-up and top-down influences according to such a Bayesian hierarchical structure. Estimation of the model on eye-movement data, collected while participants were searching for brands of laundry detergent on a simulated shopping shelf, revealed that about two thirds of the variation in the salience of products on the shelf was due to bottom-up factors and one third due to top-down factors.

Several components of these search models, such as top-down effects and salience-based and systematic processing, can be used in stand-alone models applied in experiments in which other uncertainties have been strictly controlled for. For example, the relative salience of an object fixated during search can be estimated as the average fixation-weighted Euclidean distance between that object and the target object in the CIELAB color space (van der Lans et al., 2021). Such an operationalization of salience encapsulates the extent to which participants fixate on distractors that are similar to the target, with shared color being a dominant source of visual similarity (Rutishauser & Koch, 2007). Results across a range of products (sunglasses, sneakers, perfumes) revealed that search efficiency gains occur via top-down suppression of perceptual features of competing, *distractor* products rather than by enhancement of the features of the target product (van der Lans et al., 2021; Fig. 2, P5).

Recent work has captured the time-dependency of eye movements during search and related perceptual decision tasks via first-order autoregressive (AR) model formulations, rather than via HMMs. In AR models, the observed variable (fixations) depends probabilistically on its own past values. AR formulations have been used to capture time-dependencies in eye movements in Generalized Linear Mixed models by Cho, Brown-Schmidt and Lee (2018), and in tree-based Item Response Theory models by Cho, Brown-Schmidt, Boeck and Shen (2020). These models describe the POR (point of regard) of the eyes during search as a sequence of binary micro-decisions. The neuro-psychological support for such formulations, although not used to motivate the original modeling, may come from such phenomena as inhibition of return (IOR; Castel, Pratt, & Craik, 2003), attentional momentum (Pratt, Spalek, & Bradshaw, 1999), and systematic oculomotor tendencies (Spalek & Hammad, 2005; Fig. 2, P5). Looking at time dependencies via

a hierarchical Bayes logistic regression, Shi and Trusov (2021) study how people explore search engine (Google) results. They investigate what part of the results page people look at, whether they scroll, which items they look at, and which item they click on. The results show that the prior scan-path affects these micro-decisions, along with top-down effects of search goals and bottom-up effects of the semantic context and the spatial layout of the page.

3.2. Eye-Movement Models of Specification Search

Research in agricultural and food economics has examined specification search for price, nutrition, sustainability and other attributes of products, packaging, and food labels, as reviewed by Van Loo et al. (2018). The purpose of that research is to understand which perceptual features and abstract attributes of objects influence perceptual ("find the most sustainable products on display") and preferential decisions ("choose the products that fit your preferences best"). Research in this tradition has explored bottom-up and top-down factors, the latter including for example having "no goal" versus having "general health" or more specific "reduce sodium intake" goals (Oswald, Adhikari, & Mohan, 2022). For instance, van Herpen and van Trijp (2011) found that pictorial "traffic light" labels improved search for attributes of food more than textual labels did, even when people were not under time pressure. These studies provide descriptions of specification search as it unfolds in daily life.

Eye movements during specification search have also been studied as part of other than pure specification tasks (see Table 1). For example, research has examined visual exploration of so-called retail feature advertisements, which display multiple (price discounted or otherwise featured) products in a single advertisement. Visual exploration of these ads is a mixed target/specification search task (Table 1), during which people aim to reduce uncertainty about whether a preferred product is present and if so, what its attributes are. Pieters, Wedel and Zhang (2007) examined such a task with a hierarchical Bayes model to capture the multilevel nature of multiple ads nested in such ad displays. They used measures of the visual distinctiveness of the target (perceptual difference between a target ad and its competitors in terms of the sizes of their ROIs -regions of interest) and the heterogeneity of distractors (perceptual differences among the other, competing ads) as predictor variables. Their results showed, in support of fundamental psychological theories (Duncan & Humphreys, 1989), that target distinctiveness facilitated and distractor heterogeneity hampered search. The modelling results enabled optimizing the organization of the ad display to improve search performance for each of the advertised products. Strategy switching during specification search tasks (Liechty et al., 2003) has also been studied, for instance during exploration of theme advertisements (Liechty et al., 2003; Wedel et al., 2008), word-sentence processing (Simola, Salojärvi, & Kojo, 2008), picture viewing (Haji-Abolhassani & Clark, 2013, 2014), and tracking moving targets (Kim, Singh, Thiessen, & Fisher, 2020).

In view of the ubiquity of search and choice behavior by consumers on attribute-by-product displays and the potential costs of errors, the paucity of eye-movement research in this domain is surprising. But, for instance, tracking eye movements on "balanced score cards" that managers rely on to assess employees (Chen, Jermias, & Panggabean, 2016) revealed factors that support better accounting decisions. Eye-movement research into search and choice on row-column displays has led to qualitatively different insights than obtained by more traditional process tracing methodologies. Research relying on traditional process tracing methodologies, such as information display boards (e.g., Lohse & Johnson, 1996) or one-way mirrors to observe people's eye movements during decision making (e.g., Russo & Leclerc, 1994) initially identified two or three well-ordered sequential stages in preferential decision making (from screening to evaluation, and then to verification). Likewise, eye-tracking studies initially found that in repeated conjoint choice tasks, participants switch from using attribute-based strategies first to product-based strategies later on (Meißner & Decker, 2010). Research with Hidden Markov Models (HMMs) has amended this

orderly, sequential, two or three-stage perspective on search and choice processes. Shi et al. (2013) used a three-layer Hierarchical Hidden Markov Model (HHMM) to represent eye movements made to acquire information on attribute-by-product matrices. Their model has two hierarchically connected layers of unobserved states, where the transitions between states in each layer are governed by a Markov process. Given the states of the first hidden layer, the output layer describes the eye movements using Markov transition probabilities between ROIs that are defined via a row (attribute)-column (product) spatial grid. The first hidden layer consists of two states that represent the latent by-attribute or by-product information acquisition strategies, given the states of the second layer. That second layer contains states that allow for switching between these first-layer strategies. The second-layer states, the authors speculated, might reflect top-down processes that activate a specific information-acquisition strategy. Notably, findings from the HHMM applied to data about choices between laptops presented on a 12 by 4 row-column display are at variance with some of the received knowledge obtained from traditional process tracing methodologies and descriptive modeling. First, rather than starting with attribute-based search, people tend to start and end with product-based information search, while in between relying on attribute-based search. Second, rather than switching only once or twice between attribute-based and productbased search, people switched many times. Third, the switching patterns between attribute- and product-based processing depended on the orientation of the display (attributes-by-products or products-by-attributes).

In another application, Chuk et al. (2020) use a similar two-level HHMM to describe visual information acquisition during choice of one out of two faces. The purpose of the analysis was to identify regions of interest (ROIs) that are unknown in terms of their number, locations, and boundaries, and the saccade patterns between them. The states in the first hidden layer are interpreted as individual-specific ROIs, that is regions on the human face that a participant uses in deciding which face they find more appealing. The second layer of the HHMM captures the transitions between states of the first hidden layer. The fixation locations at the output layer are modelled via a spatial Gaussian distribution. The HHMM was estimated for each individual separately using a (variational) Bayesian approach that determines the optimal number of states automatically. The results reveal two dominant patterns of eye movements. In the first pattern, people fixate on the preferred face sooner, but the accuracy of inferring their final choice from the eye movements is lower. In the second, people transition from exploration to fixating on the preferred face later, but the tendency to fixate on their preferred face is stronger.

4. Preferential Decision Making: Eye-Movement Models of Choice

Models of preferential or value-based decision making in marketing and economics traditionally rely on the assumption that people integrate all available information on attributes of the alternatives to arrive at the value or utility of each choice option (Fig. 2, I3). For instance, in sequential search model in economics (Weitzman, 1979), people are assumed to first sort choice alternatives in order of expected marginal utility and then sequentially search the options until the expected utility of (examining) the next alternative is less than the current alternative (Moorthy et al., 1997). During that type of product-based processing, people are assumed to rely on compensatory (Fishbein & Ajzen, 1975; Von Neumann & Morgenstern, 1947) preference models of a linear additive (subjective expected utility) form. Then, the choice outcome corresponds probabilistically to the alternative with maximum utility among the set of all alternatives. Models of such preferential decisions involve the Multinomial Logit or Probit functional forms, which link utility to decision outcomes (Fig. 2, O3). The Mixed Multinomial Logit model is an extension that accommodates unobserved heterogeneity by allowing its parameters to follow a Normal distribution across individuals.

These assumed processes occur during integration tasks (Table 1), but integration does not necessarily involve utility maximization, nor full information, nor linear compensatory choice rules (McFadden, 1974). In real-life decision making, people use various simplifying strategies and choice heuristics to prevent cognitive overload and to balance search benefits and costs (Kahneman, 2011). For instance, choice inertia and habitual choice are common for (frequent) low risk, low-involvement decisions. These phenomena can be readily accommodated in choice models through Markov formulations that include the immediate past decision outcome or via an exponentially smoothed average of past decision outcomes (Guadagni & Little, 1983; Keane, 1997). These formulations in essence capture reinforcement learning where the learning rates are prespecified (equal to one) or estimated (Rescorla & Wagner, 1972).

Reflective, non-habitual, decision making is more prevalent for high-involvement products and contexts. But even there, some information, such as specific attribute-alternative combinations or even entire attributes or alternatives, may not be considered at all, because the expected costs of processing outweigh the expected benefits (Sims, 2003). Thus, to lower cognitive effort, people use simplifying heuristics that are based on only a part of the available information (Payne, Bettman, & Johnson, 1988; 1992). Examples include satisficing (Simon, 1955), lexicographic (Von Neumann & Morgenstern, 1947), and elimination-by-aspects rules (Tversky, 1972). To capture these "boundedly rational" decisions, selection mechanisms have been introduced in choice models to reflect how people narrow down the choice set prior to making a full-information choice. Twostage nested logit models (McFadden, 1981), joint models of consideration and choice (Roberts & Lattin, 1997), and models that account for decision strategy shifts (Swait & Adamowicz, 2001) are examples. Alternatively, latent binary selection indicators have been included into utility functions as moderators of product attributes, to capture a variety of decision heuristics (Gilbride & Allenby, 2004, 2006; Jedidi & Kohli, 2008). Research incorporating eye-movement data in choice models has revealed that people rely on compensatory (maximizing) decision making in some instances (Glöckner & Herbold, 2011), on satisficing rules that involve people attending to alternatives until a satisfactory one is found in other (low-involvement) instances (Stüttgen et al., 2012), or on hybrid decision strategies that involve a combination of optimal search and satisficing (Reutskaja, Nagel, Camerer, & Rangel, 2011).

To structure the burgeoning field of eye-movement research on decision making, we distinguish three categories of approaches, based on whether eye movements are (1) used as explanatory variables to infer what attributes and choice options (products) are processed (Sect. 4.1), (2) used in psychological models as indicators that moderate evidence accumulation (Sect. 4.2), or (3) modeled endogenously along with choice outcomes based on economic principles (Sect. 4.3) (details of the key models are provided in Appendix 2).

4.1. Preferential Decision Models with Eye-Movements as Explanatory Variables

Eye movements have been used as explanatory variables in choice models to capture the information that people use or ignore in making decisions. *Inattentional blindness* is an automatic phenomenon that occurs when certain objects or attributes of a scene are not noticed (even though they might receive eye fixations) because attention is devoted to another task or object (Mack & Rock, 1998). In the context of multi-attribute choice, attribute inattention (Fig. 2, I3, P5) occurs when certain attributes are not processed (Hensher, Rose, & Greene, 2005), and object inattention occurs when one or more alternatives are not considered in the choice decision (Roberts & Lattin, 1997). But such inattention can be mitigated. When choice tasks in eye-tracking studies are designed to be incentive compatible, participants utilize up to twenty percent more of the attribute information presented to them (Yang, Toubia, & de Jong, 2018). Nonetheless, to accommodate inattention, eye-movement selection indicators (yes-no fixated) have been used as explanatory variables in choice models, as direct effects (Pieters & Warlop, 1999), as moderators of the

attributes (part-worths) of products (Balcombe, Fraser, & McSorley, 2015; Yegoryan, Guhl, & Klapper, 2020), or as mediators to explain decision outcomes (Chandon et al., 2009; Pieters & Warlop, 1999).

4.2. Psychological Process Models of Preferential Decision Making

A stream of literature in psychology has developed process models of decision making that are calibrated on response times and choice outcomes. Such Sequential Sampling Models (SSM) (Ratcliff, 1978; Ratcliff, Van Zandt, & McKoon, 1999; Smith & Ratcliff, 2004) assume that evidence (perceptual choice) or utility (preferential choice) on two or more choice alternatives accumulates stochastically, until one alternative has accumulated sufficient evidence to cross a decision threshold. A larger value of the threshold results in more accurate but slower responses. Drift Diffusion Models (DDM) are a special case. In the basic DDM version, the accumulation of evidence on each alternative is described by a Wiener diffusion process (which has independent Gaussian increments). DDMs were originally developed to describe accuracy and latency of two-alternative perceptual decision tasks (Ratcliff, 1978; Smith & Ratcliff, 2004; Voss, Nagler, & Lerche, 2013), but have later also been applied to examine preferential decision making (Dutilh & Rieskamp, 2016; Summerfield & Tsetsos, 2012; Voss, Rothermund, & Voss, 2004).

Commonly, DDMs are fit to preferential decision outcomes assuming an underlying diffusion process according to which a unitary subjective value for each alternative accumulates over time. For example, DDMs have been used to describe choices between pairs of food products using preference ratings for these products provided by participants prior to the choice task as exogenous input (Milosavljevic, Malmaud, Huth, Koch, & Rangel, 2010). DDMs have been fit to choices among multiple alternatives as well, based on the assumption that the differences in accumulated evidence among them favors one of the options and determines choice. Other SSMs describe multi-alternative choice with multiple diffusion processes, each with their own drift rate, which race until the first one crosses a threshold (Usher & McClelland, 2001; Usher, Olami, & McClelland, 2002).

A critical assumption of traditional DDMs is that visual information uptake is randomly distributed across alternatives. Therefore, these models imply that eye movements are uniformly distributed over alternatives (and attributes), which is similar to the implicit assumption in compensatory decision models that people attend equally to all information in the choice task (Glöckner & Herbold, 2011; Orquin et al., 2021). Despite some challenges in generalizing DDMs, and SSMs more generally, to multi-alternative preferential choice in realistic contexts (e.g., Mormann & Russo, 2021), the models provide an important step toward connecting neuro-physiological processes to unobserved preference formation. With a few exceptions (e.g., Satomura, Wedel, & Pieters, 2014), sequential sampling models have vet received limited attention in the marketing and economics literature as process models for decision making. Nonetheless, the Poisson race model, which also rest on sequential sampling of information during stimulus exposure, has been applied to conjoint choice tasks (Ruan, MacEachern, Otter, & Dean, 2008). That model assumes that information signals arrive for each alternative according to a Poisson process (with time-varying rate), until an (integer) threshold is crossed. While that model was fit to only macro-outcomes, choice and latency, Poisson race models for perceptual decisions have used eye-tracking data to express how random chunks of information arrive via eye fixations according to a Poisson process until a threshold is crossed that determines the decision outcome (Wedel & Pieters, 2000).

Extensions of DDMs have included eye movements to describe perceptual (Rangelov & Mattingley, 2020; Tavares, Perona, & Rangel, 2017) and preferential decision-making (Callaway et al., 2021; Cavanagh, Wiecki, Kochar, & Frank, 2014; Towal, Mormann, & Koch, 2013). The key feature of the attentional DDM (aDDM) by Krajbich, Rangel and colleagues (Krajbich et al., 2010; Krajbich, Lu, Camerer, & Rangel, 2012; Krajbich & Rangel, 2011) is that the information

on any alternative accumulates only while the eyes fixate on that alternative. The parameters of the model capture the speed of value accumulation, the bias toward the fixated option, and the error variance.

The initial aDDM predicts that alternatives fixated first and fixated more frequently are more likely to be chosen. Still, it does not allow for effects of bottom-up factors (Fig. 2, I1, I2, P3), nor for effects of the accumulating utility on eye movements (Fig. 2, P2; Mormann & Russo, 2021). Thus, it assumes the eye-movement process to be exogenous to the preference formation and choice processes, that is, saccades between choice options are assumed to be random, which is at variance which evidence reviewed here. Recent extensions of the aDDM (Gluth et al., 2020) allow the accumulated value of an option to increase the likelihood that it will be fixated (Fig. 2, P2). This extended model accounts for various eye-movement patterns, including the first fixation and the evolution of fixations on choice alternatives over time. Likewise, Song, Wang, Zhang and Li (2019) have proposed a sequential sampling model in which alternatives with higher (value/utility) uncertainty are more likely to be fixated (Table 1). In their model, eye fixations deliver additional information that reduces the uncertainty about choice alternatives via a Bayesian updating mechanism. It has been shown (Callaway et al., 2021) that with Bayesian updating the selection of which alternative to fixate and when to terminate the process can be formulated as a dynamic optimization problem, in which (costly) eye movements are allocated optimally over time to acquire information that reduces uncertainty about the values of the alternatives.

4.3. Economic Models of Preferential Decision Making

Recent models of preferential decision-making aim to predict future choice between multiple alternatives while endogenizing the complete eye-movement process. For example, Martinovici et al. (2021) use not just the final fixations, as in research on the gaze cascade (Shimojo et al., 2003), or the sum of fixations as in aDDMs, but the entire trajectory of eye movements for each of the choice alternatives to predict preferential decision making in a naturalistic choice task. They develop a joint model of eye movements and choice outcomes that describes the gaze on product ROIs over time as a set of covarying higher-order polynomials. It enables prediction of the final choice from gaze patterns well before the actual choices are implemented, out-of-sample. It is a predictive model, however, that is agnostic about the underlying interplay between attention and preference formation prior to choice.

Researchers have recently applied economic modeling frameworks (Gabaix, Laibson, Moloche, & Weinberg, 2006; Moorthy et al., 1997; Weitzman, 1979) to eye movements, in order to describe information search and choice for high-involvement products (e.g., laptops, cell phones). These frameworks describe information search as a sequential process with an optimal solution. At each step in the process, a decision maker examines the alternative with the highest utility and stops when the realized utility of that alternative is larger than the expected utility of the next alternative (Ursu et al., 2021; Yang et al., 2015). Information search in these models is thus determined by the expected value that the next piece of information provides. The models in question simultaneously embed micro- (fixation) and macro- (choice) decisions in a utility maximization, multinomial logit, modeling framework. The models assume that people use Bayesian updating of their prior uncertainty about the value of an attribute level (Table 1) with a noisy chunk of information obtained by fixating a product attribute, resulting in an updated product value. The expected utility is modeled as a weighted sum of the part-utilities of the attribute levels. The utility of search (and thus implicitly its cost) is specified as a function of eye-movement characteristics, such as the saccade length and direction (Ursu et al., 2021; Yang et al., 2015) to allow for local clusters of fixations and for horizontal and vertical saccade tendencies. Unobserved heterogeneity in the models' parameters across individuals is accommodated via a Normal distribution. The results of that research show that integrating eye movements into models of search and choice

improves parameter estimates, the understanding of the choice process, as well as out-of-sample predictions.

The appeal of such economic approaches is that they (1) integrate choice and eye-movement decisions in a single framework, (2) postulate a plausible economic (cost-benefit) mechanism for eye-movement decisions (Callaway et al., 2021), and (3) assume that utility/value evolves during the decision process and affects fixations (Gluth et al., 2020; Ursu et al., 2021). However, a limitation of current implementations is that known influences of visual factors and attentional processes (Fig. 2, bottom up: I1, I2, P3, and top down: I4, P4, P5) are not accommodated or only in a basic way. Also, the assumed (one-step) forward-looking mechanism remains as yet unverified.

Research for low-involvement value-based decisions (e.g., snacks) has also relied on an utility framework, but has specified different underlying processes (Reutskaja et al., 2011; Stüttgen et al., 2012). For low-involvement decisions, target search is more likely a main component of the choice task (Table 1), with people relying on the perceptual salience of objects, habitual decision making, and satisficing choice rules. For instance, in modeling choice of one out of a large set of products under time pressure, it was assumed that people may stop and choose after having fixated all alternatives at least once or after running out of time (Reutskaja et al., 2011). This assumption is in line with a satisficing choice heuristic where search stops after the value of the currently fixated alternative exceeds a reservation value.

Stüttgen et al. (2012) extended the target search model of van der Lans et al. (2008a) by including a third state in the HMM, in addition to the localization and identification states. The model specifies a satisficing rather than maximizing process (Simon, 1955). Specifically, in the third, termination state an individual chooses a product that is satisfactory, as follows (products that are undetermined are chosen with a probability close to zero). The individual assigns the status satisfactory, unsatisfactory, or undetermined to each product at each fixation. If a product (here, instant noodles) has not been fixated, its status is undetermined. To determine whether a product is satisfactory, the individual needs to fixate all attributes of that product and the value of each attribute needs to be acceptable. The model accounts for strategy switching, where the switching probabilities depend on the status of the products and the number of prior fixations. It allows people to proceed to a new choice option before returning and making a final choice, however, which is at variance with a pure satisficing heuristic (Gilbride & Allenby, 2004, 2006; Jedidi & Kohli, 2008) or the process assumed in aDDMs (Gluth et al., 2020; Krajbich et al., 2010; Song et al., 2019).

5. Conclusions and Opportunities for Future Research

Eye movements offer a unique window on perceptual, cognitive, and evaluative processes of people who are engaged in decision making tasks. They afford tests of fundamental theories, new insights into search and choice phenomena, and predictions of future decisions. Understanding how complex decisions are made in real life requires recognizing that real-life decision tasks are mixtures of elementary tasks through which people try to reduce multiple uncertainties (Table 1). Therefore, a key component of our framework in Table 1 and Fig. 2 is the idea that people switch between strategies to reduce the multiple uncertainties inherent to complex decision tasks. In addition, the execution of these tasks is subject to a myriad of bottom-up and top-down factors that influence the perceptual, cognitive, and evaluative processes that drive observable micro-(eye movements) and macro- (choice) decision outcomes (Fig. 1). Psychometric, econometric, and statistical models of these micro- and macro-decisions have been developed for somewhat different decision-making contexts and are therefore based on different assumptions tailored to these contexts. To advance insights into search and choice, a synthesis of these models is called

for. The present review aims to provide a first step toward such a synthesis, and the research agenda that we propose next serves that purpose.

First, while existing eye-movement models of target (van der Lans et al., 2008a; 2008b; 2021) and specification search (Shi et al., 2013) capture perceptual decision tasks (Table 1) relatively accurately, there are opportunities to improve upon them. Extant models of target and specification search are disconnected and cannot directly be used to predict choice outcomes, which presents an opportunity for future research. In addition, these models have relied on a priori specified perceptual features and regions of interest (ROIs). Further research can establish whether such prespecified variables most accurately capture the visual inputs that people use to process information during perceptual decision tasks. Advances in computer vision and machine learning could be used to (automatically) extract a large variety of characteristics of images, video, and text. Furthermore, eve-movement patterns themselves can be used to extract ROIs that are relevant for the decision task at hand (Chuk et al., 2020). Those data could be used in learning models (Ursu et al., 2021; Yang et al., 2015) to better understand what information people use during identification and specification processes (see Table 1). Further, the rich body of knowledge on eye movements during reading (Rayner, 1998; Rayner, Schotter, Masson, Potter, & Treiman, 2016) may inform future studies that aim to refine models of people's information acquisition from text during decision making.

Second, there are opportunities to extend DDMs and, similarly, SSMs. These models have been used to fit preferential decisions by postulating mechanisms of how the accumulation of evidence reduces decision uncertainty, similar to those for perceptual decision making (Callaway et al., 2021; Dutilh & Rieskamp, 2016). Several directions for improved theorizing and modeling remain: (1) extant research often infers the plausibility of the underlying sequential sampling processes from the statistical fit between inputs and outputs, rather than by examining the processes directly through eye movements. (2) DDMs typically rely on the assumption that visual information uptake is uniformly distributed across alternatives and attributes (for an exception see Gluth et al., 2020). This assumption is at variance with evidence that attention tends to be directed to higher valued options or the emerging winner during choice tasks (Martinovici et al., 2021; Meißner, Musalem, & Huber, 2016; Pieters & Warlop, 1999; Shimojo et al., 2003). (3) Preferences are often assumed to be fixed and are measured prior to the decision task. The drift in a DDM is then to be interpreted as the accumulation of information on those preferences retrieved from memory (Milosavljevic et al., 2010). While this simplification may provide a reasonable description of habitual, perceptual decision making, it hampers the successful application of DDMs to high-involvement, preferential choice contexts in which perceived utility of choice options and eye movements toward them may interact. The attentional DDM takes a step forward by incorporating eye movements into the value-accumulation process (Krajbich et al., 2010), but the mechanism driving eye movements is still mostly assumed exogenous (Mormann & Russo, 2021). In addition, the assumption that all attention reflects or adds positive utility requires tests in contexts where people attend to negative outcomes to avoid. More research is thus needed on the dynamic interplay of attention and value accumulation in DDMs (Fig. 2, P1 and P2), also because that interplay causes violations of IIA decision making (Gluth et al., 2020). IIA (Independence of Irrelevant Alternatives) specifies, restrictively, that the ratio of choice probabilities between two alternatives should not depend on the presence of a third alternative (Louie, Khaw, & Glimcher, 2013; Luce, 1977). Recent work that extends the aDDM by formulating the probability that an alternative is fixated as a (logistic) function of its accumulated value (Gluth et al., 2020) holds promise. Further, research is needed to alleviate the assumption in extant DDMs that decisions are based on accumulation of a unitary subjective value, by allowing for representations of multi-attribute decision making similar to those in economic models of choice. Some DDMs for perceptual decisions already incorporate such formulations. For example, Noguchi and Stewart (2018) propose an evidence accumulation model based on observations from process-tracing and

eye-movement models in which decisions are based on the accumulation of pairwise comparisons of attributes.

Third, econometric models of search and choice assume that people are forward looking and make a tradeoff between choosing now based on the available information or using eye movements to search for more information and then making a decision (Ursu et al., 2021; Yang et al., 2015). The key assumption that people are (one-step ahead) forward looking when making mostly subconscious eye movements remains to be verified, however. That assumption might hold for high-involvement contexts involving specification search, but less so in low-involvement contexts where target search dominates preferential choice. A further limitation of current econometric applications is that several key attention processes (Fig. 2, P3 and P5) are not considered or are included via fairly restrictive assumptions. For example, visual factors, which have a large impact on choice decisions (Orquin et al., 2021), are typically absent. Extant models do not account for consecutive refixations on the same object (Chandon et al., 2009; Krajbich et al., 2010; Martinovici et al., 2021; Yang et al., 2015), which is a striking characteristic of scan-paths across a wide range of tasks (Liechty et al., 2003; Musalem, Montoya, Meißner, & Huber, 2021; Rayner, 1998; van der Lans et al., 2008b).

Fourth, despite the different approaches taken in the models developed in psychology, marketing and economics, there are also some remarkable similarities. As a case in point, econometric (random utility) choice models and some SSMs used in psychology are mathematically equivalent (Webb, 2019). Further, the threshold-crossing phenomena in SSMs resemble satisficing (Glimcher, 2010; Reutskaja et al., 2011; Stüttgen et al., 2012). DDMs are mathematically equivalent to Bayesian decision models (Bitzer, Park, Blankenburg, & Kiebel, 2014). The Bayesian updating mechanism by which fixations reduce uncertainty in process models in psychology (Callaway et al., 2021; Song et al., 2019) is almost identical to the mechanism that has been assumed in models of search and choice in marketing and economics (Ursu et al., 2021; Yang et al., 2015). Finally, the hierarchical Bayesian priors that have been used in psychology and computer vision to capture top-down influences on eye-movements (Borji et al., 2014; Torralba et al., 2006) are identical to those implemented in statistical models of search in marketing (van der Lans et al., 2008a). Such similarities in approaches developed in rather independent streams of research not only reveal convergent validity of assumptions underlying different models, but also provide touchpoints to further integrate models from psychology, economics, and marketing. This review has tried to lay a foundation for such an integration by incorporating the elementary tasks and cognitive processes involved in perceptual and preferential decision making in a single conceptual framework (Table 1; Fig. 2).

Fifth, more theoretical and empirical work on the role of tasks and task switching in search and choice is called for. We have noted that decisions in real-life contexts comprise of elementary subtasks (Table 1) that may, for example, involve both target search and specification search, or habitual and reflective decision making. Future research could integrate mathematical representations of these processes by capitalizing on representations of task/strategy switching and existing communalities between model formulations. Strategy switching has already been demonstrated for micro-decisions in a wide range of tasks (Chuk et al., 2020; Shi et al., 2013; Stüttgen et al., 2012; van der Lans et al., 2008a; 2008b; Wedel et al., 2008). Whereas dual-process theories for macro-decisions abound (Damasio, 1994; Glimcher, 2010; Kahneman, 2011), and initial evidence for extensive strategy switching during these decisions comes from eye movement recordings (Venkatraman et al., 2014), more work is awaited that incorporates strategy switching in psychological (SSM) and econometric (sequential search) models and applies these to more fully understand how people switch between decision strategies, for example based on cost-benefit tradeoffs (Payne et al., 1988; Venkatraman et al., 2014).

In closing, the study of search and choice in natural conditions requires strong theory as well as sound psychometric, econometric, and statistical models to infer the underlying perceptual, cog-

nitive and evaluative processes. Marketing and economics have provided a fruitful testing ground for theories and models of attention and decision making, because of their focus on consumer decision making toward complex pictorial and textual, static, dynamic, and multisensory stimuli in natural contexts. Eye-movement data have come to play a critical role. Naturally, therefore, research, product and service companies have embraced eye tracking as a tool. With eye tracking via regular webcams being a low-cost option, eye-movement recording is already implemented in smart phones, car navigation, and head-mounted virtual reality devices. Measures produced as a by-product of eye movement recording, such as fixation durations, micro-saccades, distance between eye and stimulus, pupil dilations, and facial expressions of emotions, could be further exploited (Pieters & Wedel, 2020). Given those developments, understanding, predicting, and improving decision making in practice would benefit greatly from adopting the models reviewed here, and the extensions and generalizations that we hope the proposed framework stimulates.

Acknowledgments

The authors are grateful to Jacob Orquin for very useful comments on this manuscript.

Open Access This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/.

Publisher's Note Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Appendix 1: Neural Foundations of Eye Movements During Search and Choice

Light is projected through the cornea and the lens of the eye, on the retina. The photoreceptors in the retina process color (cones; in the fovea), luminance (rods; in the periphery) and contrast (bipolar and ganglion cells), amongst others. Via the fovea, the eye extracts high-acuity information from a region of about 2 degrees of visual angle (Palmer, 1999). Signals from the retina are sent via the optic nerve to the Lateral Geniculate Nuclei (LGN; in the Thalamus). From there on information of the two eyes is combined and passed (via the optic radiation) to area V1 in the Occipital Lobe. The visual areas in this lobe form a hierarchy and specialize in the processing of *low-level visual features*, such as edges (V1, V2), depth (V2), contours (V2), simple shapes (V4), colors (V4), and self-(V5), and object (V6) motion. These areas are organized as retinotopic maps.

The basic perceptual features that are processed *bottom-up*, pre-attentively via areas V1-V4 (Treisman & Gelade, 1980; Wolfe, 1994) are combined to form a *salience map* (Itti & Koch, 2001; Itti et al., 1998; Koch & Ullman, 1985). The salience map is a retinotopic map that represents the conspicuity of locations in the visual field. It is represented in V1, the Superior Colliculus (SC) and the Frontal Eye Fields (FEF) (Koch & Ullman, 1985; Niebur & Koch, 1998; Treue, 2003), where its representation may be short-lived (Donk & Soesman, 2010).

These areas V1-V6 output information via two streams, the ventral ("what") and the dorsal ("where") stream (Ungerleider & Mishkin, 1982). The *ventral stream* processes mostly foveal information, connects to the Inferotemporal Cortex (ITC) and is concerned with object recognition

(Bar, 2004). The ITC projects into the Perirhinal Cortex and Hippocampus amongst others, which are involved in memory, and into the Prefrontal Cortex (PFC). The PFC is involved in planning and reflective *goal-directed decisions*, while the (dorsolateral) striatum is involved in *habitual or reflexive decisions;* both regions are thought to encode value (Daw & O'Doherty, 2014; Glimcher, 2010).

The *top-down influence* of *goals*, for example, originates in the PFC (Desimone & Duncan, 1995), while *value-driven attentional capture*, whereby stimuli associated with a reward attract attention involuntarily, involves dopamine production (which confers motivational salience) from the basal ganglia (Berridge, 2012; Hayhoe & Ballard, 2005). Top-down influences occur along the entire visual hierarchy in areas V1-V5, ITC, SC, and PPC (Corbetta & Shulman, 2002; Tipper, Weaver, & Houghton, 1994) and enhance task-relevant and suppress task-irrelevant features and objects (Sawaki & Luck, 2010).

The *dorsal stream* processes spatial information mostly from the peripheral retina and is connected to the Posterior Parietal Cortex (PPC) which is involved in the location of and planning of action toward objects and in integrating sensory information. The Retrosplenial Cortex (RSC) holds a representation of scenes based on coarse global information and activates scene schemas in the Parahippocampal Cortex (PHC; Bar, 2004). The PHC primes specific object representations in the ITC (Bar, 2004) and is thereby responsible for contextual guidance of eye movements.

The SC and FEF are involved in the production of *eye movements*. The SC receives input directly from the LGN and directs the eyes to salient locations. The FEF direct eye movements to remembered stimuli and are involved in both the facilitation of voluntary saccades and in the inhibition of reflexive saccades that originate in the SC (Rafal, Machado, Ro, & Ingle, 2000). Specific areas of the SC and FEF that are part of the dorsal stream are involved in long-amplitude saccades, whereas other areas of the SC and FEF that are part of the ventral stream produce short-amplitude saccades (Bullier et al., 1996; Thompson, 2005). The SC is also involved in the *Inhibition of Return* (IOR; Sapir, Soroker, Berger, & Henik, 1999), which temporarily inhibits the return of the eyes to previously attended locations or objects (Castel et al., 2003).

Appendix 2: Details of Selected Search and Choice Models

This appendix details some of the models described in the main text, using the following notation. At time t, a visual scene is comprised of coordinates (l_1, l_2) , with $l_1 = 1, ..., L_1$ and $l_2 = 1, ..., L_2$. Image content is given by $M_{kt}(l_1, l_2)$, k = 1, ..., K, for K visual features, and h = 1, ..., HROIs with $D_{ht}(l_1, l_2) = 1$ if location (l_1, l_2) falls into ROI h at time t, and zero otherwise. If the images are stationary, the subscript t is dropped. The vector $x_{i,t} = (l_1, l_2)$ represents the location of fixation t for individual i. The entire scan-path is denoted as $x_{i,1:T}$.

5.1. Model of Target Search

Using a spatial point process, van der Lans et al. (2008b) model the location of each fixation t for person i in image-pixel coordinates (l_1, l_2) . The explanatory variables $V_{itj}(l_{1,}, l_2)$, j = 1, ..., J are features $M_k(l_1, l_2)$ and ROIs of packages or text of m = 1, ..., M brands, with $D_m^{pack}(l_1, l_2) = 1$ and $D_m^{text}(l_1, l_2) = 1$ if location (l_1, l_2) falls into the package or text. The individual-specific effects $\theta_{i,j}$ of each explanatory variable $V_{itj}(l_{1,}, l_2)$ depend on whether that person is in the localization or in the identification state. The HMM likelihood of the entire scan-path $x_{i,1:T}$ of individual i is:

$$p(\mathbf{x}_{i,1:T}|\Theta_i) = \prod_{s=1}^2 N(\theta_{i,s} | \mu_s, \Sigma_s) \left\{ \sum_{s_2=1}^2 \dots \sum_{s_{t_i}=1}^2 \prod_{t=2}^{T_i} \pi_{s_t | s_{t-1}} p_s(\mathbf{x}_{i,t} = (l_{1,}, l_2) | \theta_{i,s}, V_{it,s}(l_{1,}, l_2)) \right\},$$

where μ_s and Σ_s are the mean and covariance matrix of the multivariate Normal heterogeneity distribution of $\theta_{i,s}$. The Markov transition probabilities between the localization and identification states are: $\pi_{s_t|s_{t-1}}$, with $s_t \in \{1, 2\}$; $\pi_{s_2|s_1} = \pi_1$ represent the initial state probabilities. The vector of explanatory variables $V_{it,s}(l_1, l_2)$ for the localization state (s=1) are, first, pixel-level features $M_k(l_{1,1}, l_2)$. Second, systematic strategies are captured via $D_{left-right(x_{i,t-1})}^{pack}(l_{1,1}, l_2)$ and $D_{top-bottom(x_{i,t-1})}^{pack}(l_{1,1}, l_2)$ which represent the ROIs to the left and right, or top and bottom, respectively, of the previously fixated package. Third, in the identification state (s=2) the tendency to re-fixate on a text or package ROI is captured via $D_{refix(x_{i,t-1})}^{pack}(l_{1,1}, l_2)$ and $D_{refix(x_{i,t-1})}^{text}(l_{1,1}, l_2)$. The probability of fixating on a location is:

$$p_{s}\left(x_{i,t}=\left(l_{1,},l_{2}\right)|\theta_{i,s}, V_{it,s}\left(l_{1,},l_{2}\right)\right)=\frac{a_{it,s}\left(l_{1},l_{2}\right)^{2}}{\sum_{l_{1},l_{2}}a_{it,s}\left(l_{1},l_{2}\right)^{2}},$$
(A.2)

with $a_{it,s}(l_1, l_2) = V_{it,s}(l_1, l_2) \theta_{i,s}$. Here, $V_{it,1}(l_1, l_2)$ contains all explanatory variables and $V_{it,2}(l_1, l_2)$ the variables that affect identification. van der Lans et al. (2008a) extend the model by including top-down influences on salience, via a between-subjects experiment with g(i) = 1, ..., G search goals of participant i as experimental conditions. The individual-level parameters $\theta_{i,1:K}$ can then be decomposed into bottom-up (μ) and top-down ($\mu_{g(i)}$) influences, according to a hierarchical structure, $\theta_{i,1:K, s=1} \sim N(\mu + \mu_{g(i)}, \Sigma_{s=1})$, with Normal priors on μ and $\mu_{g(i)}$ and a Wishart prior on $\Sigma_{s=1}$.

5.2. Model of Specification Search

Shi et al. (2013) propose a three-layer HHMM to analyze specification search, which involves three sets of Markovian transition probabilities:

- 1. The transition probabilities between higher order states, R_t , of the upper layer, $\pi_{R_t|R_{t-1}}$
- 2. The transition probabilities between the two states, S_t , of the middle layer that represent by-attribute and by-product processing, given the states of the upper layer, $\pi_{S_t|S_{t-1}}^{R_t}$.
- 3. The probabilities of eye movements between cells on the display given the states of the middle layer $P^{S_t}(x_{i,t}|x_{i,t-1})$. Here, $P^1(x_{i,t}|x_{i,t-1})$ is a function of $D_{attr}(x_{i,t-1})(x_{i,t})$, attribute ROIs, capturing a by-attribute tendency and $P^2(x_{i,t}|x_{i,t-1})$ a function of $D_{prod}(x_{i,t-1})(x_{i,t})$, product ROIs capturing a by-product tendency.

The full HHMM model likelihood is:

$$p(\mathbf{x}_{i}|\pi) = \sum_{R_{2}=1}^{3} \dots \sum_{R_{T}=1}^{3} \left(\sum_{S_{2}=1}^{2} \dots \sum_{S_{T}=1}^{2} \pi_{R_{1}} \pi_{S_{1}}^{R_{1}} P^{S_{1}}(\mathbf{x}_{i,1}) \prod_{t=2}^{T_{i}} \pi_{R_{t}|R_{t-1}} \pi_{S_{t}|S_{t-1}}^{R_{r}} P^{S_{t}}(\mathbf{x}_{i,t}|\mathbf{x}_{i,t-1}) \right).$$
(A.3)

5.3. DDM of Preferential Decision Making

The attentional DDM (aDDM) (Krajbich et al., 2010; Krajbich et al., 2012; Krajbich & Rangel, 2011) specifies that the rate of accumulation (drift) of the decision value depends on the product or price ROI that the eyes fixate on. Alternative j is selected if its accumulated value v_{ijt} is the first

to cross some threshold τ_i : Y_i= j, if $|v_{ijt}| > \tau_i$, with $|v_{ikt}| \le \tau_i \forall k = 1, ..., J \cap k \ne j$. A larger value of the threshold τ_i result in more accurate but slower responses. For a product fixation, the drift is:

$$dv = \phi(z_1 - \beta z_2)dt + \upsilon dW, \tag{A.4}$$

for a price fixation the drift is $-\phi (z_2 - \beta z_1)$. Here, z_1 and z_2 are exogenous measures of product preference and price. dW is Gaussian noise. The parameters of the model are the drift ϕ (for the second alternative the drift is $-\phi$), the bias toward the fixated option $\beta \in (0, 1)$, and error standard deviation υ .

5.4. Economic Models of Search and Choice

The model by Yang et al. (2015) is based on the assumption that consumer's uncertainty about the true value of an attribute level l of a product, $Z_{j,m}$ (with $l = 1, ..., L_m$) is reduced by extracting an (unknown) amount η of unbiased information through an eye fixation. Assuming a uniform distribution of prior beliefs across attribute levels, the posterior probabilities that attribute m of alternative j equals l (with $l = 1, ..., L_m$), given that person i has made N_{ijm} fixations on it, are:

$$p(Z_{j,m}=l) = \begin{cases} \frac{exp(\eta \cdot N_{ijm})}{L_m - 1 + exp(\eta \cdot N_{ijm})} & \text{if } l \text{ is the true level of attribute } m, \\ \frac{1}{L_m - 1 + exp(\eta \cdot N_{ijm})} & \text{otherwise.} \end{cases}$$
(A.5)

The expected choice utility, $U_{ij(c)}$, which is accrued if product j is chosen, is modeled as a probability weighted sum of the part-utilities ($\mu_{i,m,l}$) of the attribute levels:

$$U_{ij(c)} = \sum_{m=1}^{M} \sum_{l=1}^{L_m} p\left(Z_{j,m} = l\right) \mu_{i,m,l}. \tag{A.6}$$

The utility of search, $U_{i(s)}$, is specified as a linear function of the saccade length and of horizontal and vertical tendencies. Consumers face the following maximization problem:

$$\max_{q} \left\{ \max_{q=j} U_{ij(c)} \left(N_{ij}, \eta, \mu_{i,m} \right), \ \max_{m} \left(U_{i(s)} + \sum_{l=1}^{L_{m}} p\left(Z_{j,m} = l \right) \max_{j} U_{ij(c)} \left(N_{ij}^{*}, \eta, \mu_{i,m} \right) \right) \right\}.$$
(A.7)

Equation (A.7) specifies a person i's next action q as a choice between a) stopping search and choosing option j, which results in choice utility $\max_{q=j} U_{ij(c)}(N_{ij}, \eta, \mu_{i,m})$, and b) continuing search and fixating the next attribute m of alternative j, which maximizes search utility $U_{i(s)}$ plus the expected maximum utility if the individual would stop after this next fixation, which equals $\sum_{l=1}^{L_m} p(Z_{j,m}=l) \max_j U_{ij(c)}(N_{ij}^*, \eta, \mu_{I,m})$, where N_{ij}^* is the vector with the number of fixations on option j.

Zhang, Ursu and Erdem (2020) assume consumer i's prior belief about brand j's value is $N\left(\alpha_{ij,0}, \sigma_{ij,0}^2\right)$. The prior is updated with a noisy signal, $\eta_{ijm,t} \sim N\left(\mu_{ijm}, \omega_{ijm}^2\right)$ from an eye fixation on attribute m and brand j at time t, resulting in brand value $N\left(\alpha_{ij,t}, \sigma_{ij,t}^2\right)$, with:

$$\alpha_{ij,t+1} = \frac{\frac{\alpha_{ij,t}}{\sigma_{ij,t}^2} + \frac{n_{jim,t}}{\omega_{ijm}^2}}{\frac{1}{\sigma_{ij,t}^2} + \frac{1}{\omega_{ijm}^2}}, \quad \sigma_{ij,t}^2 = \frac{1}{\frac{1}{\sigma_{ij,t}^2} + \frac{1}{\omega_{ijm}^2}}.$$
 (A.8)

The prior mean $\alpha_{ij,0}$ is a linear function of brand ownership, the prior variance $\sigma_{ij,0}^2$ is a function of brand familiarity, and the variance of the signal ω_{ijm}^2 is a function of experience with attribute m. The expected choice utility of a brand equals:

$$U_{ij,t(c)} = -\exp\left(-r \cdot \alpha_{ij,t+1} + \frac{r^2}{2}\sigma_{ij,t}^2\right),\tag{A.9}$$

reflecting risk aversion with a risk coefficient r, by which a higher posterior uncertainty about a brand, $\sigma_{ij,t}^2$, decreases its utility. The inclusion of uncertainty as a function of risk aversion in equation (A.9) prioritizes eye-movements toward attributes that have received limited attention. Furthermore, like Yang et al. (2015), consumers are assumed to be one-step ahead forward looking, such that the utility derived from searching attribute m of product j is the maximum utility derived from choosing one of the alternatives at time t + 1:

$$U_{ij,t(s)} = \max \left\{ U_{ij,t+1(c)}, U_{ik,t(c)} \right\} - c_{ij}.$$
(A.10)

Equation (A.10) provides the expected value of choosing j at t+1 after an additional search of j (the first term) minus the search costs c_{ij} , or choosing any other option k at time t (the second term) minus the search costs. The latter are parameterized as a function of the vertical and horizontal saccade length, and fixations on white space.

5.5. Satisficing Model of Search and Choice

Stüttgen et al. (2012) extend the model by van der Lans et al. (2008a) to a three-state HMM of satisficing choice, where $z_{i,t}$ indicates the state of individual i at fixation t. The model assumes that a consumer assigns the status satisfactory ($S_{ij,t}$), unsatisfactory ($U_{ij,t}$), or undetermined ($D_{ij,t}$) to each product at fixation t. If a product has not been fixated, its status is undetermined. To determine whether a product is satisfactory, a consumer needs to fixate all attributes m of that product and the value of each attribute needs to be acceptable, captured by $\psi_{im} \in \{0, 1\}$, which follows a Bernoulli distribution. Thus:

$$S_{ij,t} = \prod_{m=1}^{M_i} \psi_{im}, U_{ij,t} = \max_m (1 - \psi_{im}), \text{ and } D_{ij,t} = 1 - S_{ij,t} - U_{ij,t}.$$
(A.11)

The first two states $(z_{i,t} \in \{g, l\})$ in the HMM reflect the search process. In the global state (g), the decisions to fixate an ROI h follow a multinomial logit model with probabilities that are functions of the current status $\{S_{ij,t}, U_{ij,t}\}$ of all products j and the size of ROI h. In the local state (l), the probabilities of fixating an ROI h are a function of the previously fixated ROI and

of the ROIs bordering on it (reflecting local and re-fixations) captured via a vector of indicators $L_t(h)$, in addition to a by-attribute strategy, captured via a vector of indicators $S_t(h)$. The fixation probabilities in the local ($\varphi_{l,t}$) and global states ($\varphi_{g,t}$), with $\beta_{i,\cdot,\cdot}$ individual-level parameters to be estimated, are:

$$\varphi_{g,t}(h) \propto r(h) \exp(\beta_{i,g,0} + \beta_{i,g,1} S_{ij} + \beta_{i,g,2} U_{ij})$$
(A.12)

$$\varphi_{j,t}(h) \propto \exp(\beta_{i,l,0} + \beta_{i,l,1}S_{ij,t} + \beta_{i,l,2}U_{ij,t} + \beta'_{i,l,3}L_{t}(h) + \beta'_{i,l,4}S_{t}(h))$$
(A.13)

In the termination state (w), the consumer chooses a product j that is satisfactory. Because the majority of participants continued searching after fully evaluating their final choice (which violates pure satisficing), the model assumes a verification stage to capture this. Therefore, it is possible that multiple products are satisfactory, leading to choice probabilities of satisfactory products that are equal to p $(y_{ij}=1|S_{ij,T}=1) = 1/S$, while unsatisfactory products have a choice probability that equals zero: p $(y_{ij}=1|U_{ij,T}=1) = 0$, with S a normalizing constant. Finally, at odds with the theory of satisficing, some participants selected undetermined products, which was accommodated by a probability close to zero p $(y_{ij}=1|D_{ij,T}=1) \propto \epsilon/Swith(\epsilon \sim Beta)$. Switching probabilities between the three states represented by $z_{i,t}$ are proportional to:

$$p\left(z_{i,t}\right) \propto \begin{cases} \exp\left(\lambda_{z_{i,t}0} + \lambda_{z_{i,t}1}I\left(z_{i,t-1}=1\right) + \lambda_{z_{i,t}2}S_{if_{i,t-1},t-1} + \lambda_{z_{i,t}3}U_{if_{i,t-1},t-1}\right) \text{ if } z_{i,t} \in \{1,2\} \\ \exp\left(\lambda_{z_{i,t}0} + \lambda_{z_{i,t}1}I\left(z_{i,t-1}=1\right) + \lambda_{z_{i,t}2}N_{i,t-1} + \lambda_{z_{i,t}3}\max_{j}(S_{ij,t-1})\right) \text{ if } z_{i,t} = 3 \end{cases}$$
(A.14)

In equation (A.14), $\lambda_{z_{i,t}0}$ captures the baseline switching probability to state $z_{i,t}$. $\lambda_{z_{i,t}1}$ captures the dependence of that switching probability on the previous state $z_{i,t-1}$. Further, switching to the global and local states depends on whether the previously fixated product $f_{i,t-1}$ is satisfactory or unsatisfactory, while switching to the termination states depends on the number of fixations $N_{i,t-1}$, and whether any of the choice alternatives is currently determined satisfactory.

References

- Anderson, B. A., Laurent, P. A., & Yantis, S. (2011). Value-driven attentional capture. Proceedings of the National Academy of Sciences of the United States of America, 108(25), 10367–10371.
- Anderson, B. A., & Yantis, S. (2013). Persistence of value-driven attentional capture. Journal of Experimental Psychology: Human Perception and Performance, 39(1), 6–9.
- Anderson, N. H., & Shanteau, J. C. (1970). Information integration in risky decision making. *Journal of Experimental Psychology*, 84(3), 441–451.
- Aribarg, A., Pieters, R., & Wedel, M. (2010). Raising the BAR: Bias adjustment of recognition tests in advertising. *Journal of Marketing Research*, 47(3), 387–400.
- Atalay, A. S., Bodur, H. O., & Rasolofoarison, D. (2012). Shining in the Center: Central Gaze Cascade Effect on Product Choice. *Journal of Consumer Research*, 39(4), 848–866.
- Baddeley, R. J., & Tatler, B. W. (2006). High frequency edges (but not contrast) predict where we fixate: A Bayesian system identification analysis. *Vision Research*, 46(18), 2824–2833.
- Balcombe, K., Fraser, I., & McSorley, E. (2015). Visual attention and attribute attendance in multi-attribute choice experiments. *Journal of Applied Econometrics*, 30(3), 447–467.

Bar, M. (2004). Visual objects in context. Nature Reviews Neuroscience, 5(8), 617-629.

Berlyne, D. E. (1971). Aesthetics and psychobiology. Appleton-Century-Crofts.

Berridge, K. C. (2012). From prediction error to incentive salience: Mesolimbic computation of reward motivation. *European Journal of Neuroscience*, 35(7), 1124–1143.

- Bhatnagar, R., & Orquin, J. L. (2021). A meta-analysis on the effect of visual attention on choice. *Working paper, Aarhus University, Denmark.*
- Bialkova, S., & van Trijp, H. C. M. (2011). An efficient methodology for assessing attention to and effect of nutrition information displayed front-of-pack. *Food Quality and Preference*, 22(6), 592–601.

Bettman, J. R., Luce, M. F., & Payne, J. W. (1998). Constructive consumer choice processes. Journal of Consumer Research, 25, 187–217.

- Bitzer, S., Park, H., Blankenburg, F., & Kiebel, S. (2014). Perceptual decision making: drift-diffusion model is equivalent to a Bayesian model [Original Research]. Frontiers in Human Neuroscience, 8, 102.
- Borji, A., & Itti, L. (2014). Defending Yarbus: Eye movements reveal observers' task. Journal of Vision, 14(3), 29-29.
- Borji, A., Sihite, D. N., & Itti, L. (2014). What/where to look next? Modeling top-down visual attention in complex interactive environments. *IEEE Transactions on Systems, Man, and Cybernetics: Systems, 44*(5), 523–538.
- Bullier, J., Schall, J. D., & Morel, A. (1996). Functional streams in occipito-frontal connections in the monkey. *Behavioural Brain Research*, 76(1–2), 89–97.
- Callaway, F., Rangel, A., & Griffiths, T. L. (2021). Fixation patterns in simple choice reflect optimal information sampling. PLOS Computational Biology, 17(3), e1008863.
- Castel, A. D., Pratt, J., & Craik, F. I. M. (2003). The role of spatial working memory in inhibition of return: Evidence from divided attention tasks. *Perception & Psychophysics*, 65(6), 970–981.
- Cavanagh, J. F., Wiecki, T. V., Kochar, A., & Frank, M. J. (2014). Eye tracking and pupillometry are indicators of dissociable latent decision processes. *Journal of experimental psychology. General*, 143(4), 1476–1488.
- Chandon, P., Hutchinson, J. W., Bradlow, E. T., & Young, S. H. (2009). Does in-store marketing work? Effects of the number and position of shelf facings on brand attention and evaluation at the point of purchase. *Journal of Marketing*, 73(6), 1–17.
- Chen, M., Burke, R. R., Hui, S. K., & Leykin, A. (2021). Understanding lateral and vertical biases in consumer attention: An in-store ambulatory eye-tracking study. *Journal of Marketing Research*, 58(6), 1120–1141.
- Chen, Y., Jermias, J., & Panggabean, T. (2016). The role of visual attention in the managerial judgment of balancedscorecard performance evaluation: Insights from using an eye-tracking device. *Journal of Accounting Research*, 54(1), 113–146.
- Cho, S.-J., Brown-Schmidt, S., Boeck, P. D., & Shen, J. (2020). Modeling intensive polytomous time-series eye-tracking data: A dynamic tree-based item response model. *Psychometrika*, 85(1), 154–184.
- Cho, S.-J., Brown-Schmidt, S., & Lee, W.-Y. (2018). Autoregressive generalized linear mixed effect models with crossed random effects: An application to intensive binary time series eye-tracking data. *Psychometrika*, 83(3), 751–771.
- Chuk, T., Chan, A. B., Shimojo, S., & Hsiao, J. H. (2020). Eye movement analysis with switching hidden Markov models. Behavior Research Methods, 52(3), 1026–1043.
- Corbetta, M., & Shulman, G. L. (2002). Control of goal-directed and stimulus-driven attention in the brain. Nature Reviews Neuroscience, 3(3), 201–215.
- Damasio, A. (1994). Descartes' Error: Emotion, Reason, and the Human Brain. Grosset/Putnam.
- Daw, N. D., & O'Doherty, J. P. (2014). Multiple Systems for Value Learning. In P. W. Glimcher & E. Fehr (Eds.), *Neuroeconomics* (2nd ed., pp. 393–410). Academic Press.
- Della Libera, C., & Chelazzi, L. (2006). Visual selective attention and the effects of monetary rewards. Psychological Science 17(3), 222–227.
- Deng, X., Kahn, B. E., Unnava, H. R., & Lee, H. (2016). A "Wide" variety: Effects of horizontal versus vertical display on assortment processing, perceived variety, and choice. *Journal of Marketing Research*, 53(5), 682–698.
- Desimone, R., & Duncan, J. (1995). Neural mechanisms of selective visual attention. Annual Review of Neuroscience, 18(1), 193–222.
- Donk, M., & Soesman, L. (2010). Salience is only briefly represented: Evidence from probe-detection performance. Journal of Experimental Psychology: Human Perception and Performance, 36(2), 286–302.
- Droll, J. A., & Hayhoe, M. M. (2007). Trade-offs betrween gaze and working memory use. Journal of Experimental Psychology: Human Perception and Performance, 33(6), 1352–1365.
- Duchowski, A. T. (2003). Eye tracking methodology: Theory and practice. Springer-Verlag.
- Duncan, J., & Humphreys, G. W. (1989). Visual search and stimulus similarity. Psychological Review, 96(3), 433-458.
- Dutilh, G., & Rieskamp, J. (2016). Comparing perceptual and preferential decision making. *Psychonomic Bulletin & Review*, 23(3), 723–737.
- Einhäuser, W., Rutishauser, U., & Koch, C. (2008). Task-demands can immediately reverse the effects of sensory-driven saliency in complex visual stimuli. *Journal of Vision*, 8(2), 2.1-19.
- Fiedler, S., & Glöckner, A. (2012). The dynamics of decision making in risky choice: An eye-tracking analysis. Frontiers in Psychology, 3, 335–335.
- Findlay, J. M. (2005). Covert attention and saccadic eye movements. In L. Itti, G. Rees, & J. K. Tsotsos (Eds.), Neurobiology of attention (pp. 114–116). Elsevier Academic Press.
- Findlay, J. M., & Gilchrist, I. D. (1998). Eye guidance and visual search. In G. Underwood (Ed.), Eye guidance in reading and scene perception (pp. 295–312). Elsevier Science Ltd.
- Findlay, J. M., & Gilchrist, I. D. (2003). Active Vision: The Psychology of Looking and Seeing. Oxford University Press. Fishbein, M., & Ajzen, I. (1975). Belief, attitude, intention, and behavior: An introduction to theory and research. Addison-Wesley.
- Fisher, G. (2021). Intertemporal choices are causally influenced by fluctuations in visual attention. *Management Science*, 67(8), 4961–4981.
- Foulsham, T., & Underwood, G. (2008). What can saliency models predict about eye movements? Spatial and sequential aspects of fixations during encoding and recognition. *Journal of Vision*,82 (6), 1–17.
- Gabaix, X., Laibson, D., Moloche, G., & Weinberg, S. (2006). Costly information acquisition: Experimental analysis of a boundedly rational model. *American Economic Review*, 96(4), 1043–1068.
- Gaspelin, N., & Luck, S. J. (2018). Distinguishing among potential mechanisms of singleton suppression. Journal of Experimental Psychology: Human Perception and Performance, 44(4), 626–644.
- Gilbert, S. J., & Shallice, T. (2002). Task switching: A PDP model. Cogn Psychol, 44(3), 297–337.

- Gilbride, T. J., & Allenby, G. M. (2004). A choice model with conjunctive, disjunctive, and compensatory screening rules. *Marketing Science*, 23(3), 391–406.
- Gilbride, T. J., & Allenby, G. M. (2006). Estimating heterogeneous EBA and economic screening rule choice models. *Marketing Science*, 25(5), 494–509.
- Glaholt, M. G., & Reingold, E. M. (2009). Stimulus exposure and gaze bias: A further test of the gaze cascade model. Attention, Perception, & Psychophysics, 71(3), 445–450.

Glaholt, M. G., & Reingold, E. M. (2011). Eye movement monitoring as a process tracing methodology in decision making research. *Journal of Neuroscience, Psychology, and Economics, 4*(2), 125–146.

Glaholt, M. G., Wu, M.-C., & Reingold, E. M. (2010). Evidence for top-down control of eye movements during visual decision making. *Journal of Vision*, 10(5), 10–15.

Glimcher, P. W. (2010). Foundations of neuroeconomic analysis. Oxford University Press.

- Glöckner, A., & Herbold, A.-K. (2011). An eye-tracking study on information processing in risky decisions: Evidence for compensatory strategies based on automatic processes. *Journal of Behavioral Decision Making*, 24(1), 71–98.
- Gluth, S., Kern, N., Kortmann, M., & Vitali, C. L. (2020). Value-based attention but not divisive normalization influences decisions with multiple alternatives. *Nature Human Behaviour*, 4(6), 634–645.
- Gould, J. D. (1973). Eye movements during visual search and memory search. *Journal of Experimental Psychology*, 98(1), 184–195.
- Guadagni, P. M., & Little, J. D. C. (1983). A logit model of brand choice calibrated on scanner data. *Marketing Science*, 2(3), 203–238.
- Gutzwiller, R. S., Wickens, C. D., & Clegg, B. A. (2019). The role of reward and effort over time in task switching. *Theoretical Issues in Ergonomics Science*, 20(2), 196–214.
- Haji-Abolhassani, A., & Clark, J. J. (2013). A computational model for task inference in visual search. Journal of Vision, 13(3), 29.
- Haji-Abolhassani, A., & Clark, J. J. (2014). An inverse Yarbus process: Predicting observers' task from eye movement patterns. Vision Research, 103, 127–142.
- Hayhoe, M., & Ballard, D. (2005). Eye movements in natural behavior. TRENDS in Cognitive Sciences, 9(4), 188-194.
- Henderson, J. M., & Hollingworth, A. (1998). Eye movements during scene viewing: an overview. In G. Underwood (Ed.), Eye guidance in reading and scene perception (pp. 269–293). Elsevier.
- Hensher, D. A., Rose, J., & Greene, W. H. (2005). The implications on willingness to pay of respondents ignoring specific attributes. *Transportation*, 32(3), 203–222.
- Ho, G., Scialfa, C. T., Caird, J. K., & Graw, T. (2001). Visual Search for traffic signs: The effects of clutter, luminance, and aging. *Human Factors*, 43(3), 194–207.
- Holmqvist, K., Nyström, M., Andersson, R., Dewhurst, R., Jarodzka, H., & van de Weijer, J. (2011). Eye tracking: A comprehensive guide to methods and measures. Oxford University Press.
- Isham, E. A., & Geng, J. J. (2013). Looking time predicts choice but not aesthetic value. PLoS One, 8(8), e71698–e71698.
- Itti, L., & Koch, C. (2001). Computational modelling of visual attention. Nature Reviews Neuroscience, 2(3), 194–203.
- Itti, L., Koch, C., & Niebur, E. (1998). A model of saliency-based visual attention for rapid scene analysis. *IEEE Transactions on Pattern Analysis and Machine Intelligence*, 20(11), 1254–1259.
- James, W. (1890). The principles of psychology. Henry Holt.
- Jedidi, K., & Kohli, R. (2008). Inferring latent class lexicographic rules from choice data. Journal of Mathematical Psychology, 52(4), 241–249.
- Kahneman, D. (2011). Thinking, fast and slow. Straus and Giroux: Farrar.
- Kardan, O., Berman, M. G., Yourganov, G., Schmidt, J., & Henderson, J. M. (2015). Classifying mental states from eye movements during scene viewing. *Journal of Experimental Psychology: Human Perception and Performance*, 41(6), 1502–1514.
- Keane, M. P. (1997). Modeling heterogeneity and state dependence in consumer choice behavior. Journal of Business & Economic Statistics, 15(3), 310–327.
- Kiesel, A., Steinhauser, M., Wendt, M., Falkenstein, M., Jost, K., Philipp, A. M., & Koch, I. (2010). Control and interference in task switching-a review. *Psychol Bull*, 136(5), 849–874.
- Kim, J., Singh, S., Thiessen, E. D., & Fisher, A. V. (2020). A hidden Markov model for analyzing eye-tracking of moving objects. *Behavior Research Methods*, 52(3), 1225–1243.
- Klein, R. M. (2000). Inhibition of return. TRENDS in Cognitive Sciences, 4(4), 138–147.
- Koch, C., & Ullman, S. (1985). Shifts in selective visual attention: Towards the underlying neural circuitry. *Human Neurobiology*, 4(4), 219–227.
- Kowler, E. (2011). Eye movements: The past 25 years. Vision Research, 51(13), 1457-1483.

Krajbich, I., Armel, C., & Rangel, A. (2010). Visual fixations and the computation and comparison of value in simple choice. *Nature Neuroscience*, 13(10), 1292–1298.

- Krajbich, I., Lu, D., Camerer, C., & Rangel, A. (2012). The attentional drift-diffusion model extends to simple purchasing decisions [Original Research]. Frontiers in Psychology, 3(193).
- Krajbich, I., & Rangel, A. (2011). Multialternative drift-diffusion model predicts the relationship between visual fixations and choice in value-based decisions. *Proceedings of the National Academy of Sciences*, 108(33), 13852.
- Kruglanski, A. W., Chernikova, M., Babush, M., Dugas, M., & Schumpe, B. M. (2015). The architecture of goal systems: Multifinality, equifinality, and counterfinality in means-end relations. In Advances in motivation science (Vol. 2, pp. 69–98). Elsevier.
- Krupinski, E. A., Berger, W. G., Dallas, W. J., & Roehrig, H. (2003). Searching for nodules: What features attract attention and influence detection? *Academic Radiology*, 10(8), 861–868.

Land, M. (2019). Eye movements in man and other animals. Vision Research, 162, 1-7.

- Lee, M. D., & Webb, M. R. (2005). Modeling individual differences in cognition. Psychonomic Bulletin & Review, 12(4), 605–621.
- Lee, T. S., & Mumford, D. (2003). Hierarchical Bayesian inference in the visual cortex. Journal of the Optical Society of America, 20(7), 1434–1448.
- Liechty, J., Pieters, R., & Wedel, M. (2003). Global and local covert visual attention: Evidence from a Bayesian hidden Markov model. *Psychometrika*, 68(4), 519–541.
- Locke, E. A., & Latham, G. P. (2002). Building a practically useful theory of goal setting and task motivation: A 35-year odyssey. American Psychologist, 57(9), 705–717.
- Logan, G. D., & Gordon, R. D. (2001). Executive control of visual attention in dual-task situations. *Psychological Review*, 108(2), 393–434.
- Lohse, G. L., & Johnson, E. J. (1996). A comparison of two process tracing methods on choice tasks. Organizational Behavior and Human Decision Processes, 68(1), 28–43.
- Louie, K., Khaw, M. W., & Glimcher, P. W. (2013). Normalization is a general neural mechanism for context-dependent decision making. *Proceedings of the National Academy of Sciences*, 110(15), 6139–6144.
- Luce, R. D. (1977). The choice axiom after twenty years. Journal of Mathematical Psychology, 15(3), 215–233.
- Mack, A., & Rock, I. (1998). Inattentional blindness. The MIT Press.
- Mack, M. L., Gauthier, I., Sadr, J., & Palmeri, T. J. (2008). Object detection and basic-level categorization: Sometimes you know it is there before you know what it is. *Psychonomic Bulletin & Review*, 15(1), 28–35.
- Mannan, S., Ruddock, K. H., & Wooding, D. S. (1995). Automatic control of saccadic eye movements made in visual inspection of briefly presented 2-D images. *Spatial Vision*, 9(3), 363–386.
- Martinovici, A., Pieters, R., & Erdem, T. (2021). Attention trajectories predict brand choice. Working Paper, RSM Erasmus University.
- Mayr, U., Kuhns, D., & Rieter, M. (2013). Eye movements reveal dynamics of task control. *Journal of Experimental Psychology: General*, 142(2), 489–509.
- McCarley, J. S., Kramer, A. F., Wickens, C. D., Vidoni, E. D., & Boot, W. R. (2004). Visual skills in airport-security screening. *Psychological Science*, 15(5), 302–306.
- McFadden, D. (1974). Conditional logit analysis of qualitative choice behavior. In P. Zarembka (Ed.), Frontiers in Econometrics (pp. 105–142). Academic Press Inc.
- McFadden, D. (1981). Econometric models of probabilistic choice. In C. F. Manski & D. McFadden (Eds.), Structural analysis of discrete data and econometric applications (pp. 198–272). The MIT Press.
- Meißner, M., & Decker, R. (2010). Eye-tracking information processing in choice-based conjoint analysis. *International Journal of Market Research*, 52(5), 591–610.
- Meißner, M., Musalem, A., & Huber, J. (2016). Eye tracking reveals processes that enable conjoint choices to become increasingly efficient with practice. *Journal of Marketing Research*, 53(1), 1–17.
- Milosavljevic, M., Malmaud, J., Huth, A., Koch, C., & Rangel, A. (2010). The drift diffusion model can account for the accuracy and reaction time of value-based choices under high and low time pressure. *Judgment and Decision Making*, 5(6), 437–449.
- Milosavljevic, M., Navalpakkam, V., Koch, C., & Rangel, A. (2012). Relative visual saliency differences induce sizable bias in consumer choice. *Journal of Consumer Psychology*, 22(1), 67–74.
- Monk, T. H. (1984). Search. In J. S. Warm (Ed.), Sustained Attention in Human Performance (pp. 293-321). Wiley.
- Moorthy, S., Ratchford, B. T., & Talukdar, D. (1997). Consumer information search revisited: Theory and empirical analysis. *Journal of Consumer Research*, 23(4), 263–277.
- Mormann, M., & Russo, J. E. (2021). Does attention increase the value of choice alternatives? TRENDS in Cognitive Sciences, 25(4), 305–315.
- Musalem, A., Montoya, R., Meißner, M., & Huber, J. (2021). Components of attentional effort for repeated tasks. *Journal of Behavioral Decision Making*, 34(1), 99–115.
- Najemnik, J., & Geisler, W. S. (2005). Optimal eye movement strategies in visual search. Nature, 434(7031), 387-391.
- Niebur, E., & Koch, C. (1998). Computational architectures for attention. In R. Parasuraman (Ed.), *The attentive brain* (pp. 163–186). The MIT Press.
- Nisbett, R. E., & Wilson, T. D. (1977). Telling more than we can know: Verbal reports on mental processes. *Psychological Review*, 84(3), 231–259.
- Noguchi, T., & Stewart, N. (2018). Multialternative decision by sampling: A model of decision making constrained by process data. *Psychological Review*, 125(4), 512–544.
- Nummenmaa, L., Hyönä, J., & Calvo, M. G. (2006). Eye movement assessment of selective attentional capture by emotional pictures. *Emotion*, 6(2), 257–268.
- Oliva, A., & Schyns, P. G. (2000). Diagnostic colors mediate scene recognition. Cognitive Psychology, 41(2), 176–210.
- Oliva, A., & Torralba, A. (2006). Building the gist of a scene: The role of global image features in recognition. *Progress in Brain Research*, 155, 23–36.
- Olivers, C. N. L., Meijer, F., & Theeuwes, J. (2006). Feature-based memory-driven attentional capture: Visual working memory content affects visual attention. Journal of experimental psychology. *Human Perception and Performance*, 325(5), 1243–1265.
- Orquin, J. L., Lahm, E. S., & Stojić, H. (2021). The visual environment and attention in decision making. Psychological Bulletin, 147(6), 597–617.
- Orquin, J. L., & Loose, S. M. (2013). Attention and choice: A review on eye movements in decision making. Acta Psychologica, 144(1), 190–205.

- Oswald, C., Adhikari, K., & Mohan, A. (2022). Effect of front-of-package labels on consumer product evaluation and preferences. *Current Research in Food Science*, *5*, 131–140.
- Palmer, S. E. (1999). Vision science: Photons to phenomology. The MIT Press.
- Pannasch, S., & Velichkovsky, B. M. (2009). Distractor effect and saccade amplitudes: Further evidence on different modes of processing in free exploration of visual images. *Visual Cognition*, 17(6–7), 1109–1131.
- Parkhurst, D., Law, K., & Niebur, E. (2002). Modeling the role of salience in the allocation of overt visual attention. Vision Research, 42(1), 107–123.
- Parkhurst, D. J., & Niebur, E. (2003). Scene content selected by active vision. Spatial Vision, 16(2), 125–154.
- Payne, J. W., Bettman, J. R., & Johnson, E. J. (1988). Adaptive strategy selection in decision making. *Journal of Experimental Psychology: Learning, Memory, and Cognition*, 14(3), 534–552.
- Payne, J. W., Bettman, J. R., & Johnson, E. J. (1992). Behavioral decision research: A constructive processing perspective. Annual Review of Psychology, 43(1), 87–131.
- Payne, J. W., Bettman, J. R., & Johnson, E. J. (1993). The adaptive decision maker. Cambridge University Press.
- Pieters, R., & Warlop, L. (1999). Visual attention during brand choice: The impact of time pressure and task motivation. International Journal of Research in Marketing, 16(1), 1–16.
- Pieters, R., & Wedel, M. (2007). Goal control of visual attention to advertising: The Yarbus implication. Journal of Consumer Research, 34(2), 224–233.
- Pieters, R., & Wedel, M. (2012). Ad Gist: Ad communication in a single eye fixation. Marketing Science, 31(1), 59-73.
- Pieters, R., & Wedel, M. (2020). Heads up: Head movements during ad exposure respond to consumer goals and predict brand memory. *Journal of Business Research*, 111, 281–289.
- Pieters, R., Wedel, M., & Zhang, J. (2007). Optimal feature advertising design under competitive clutter. *Management Science*, 53(11), 1815–1828.
- Ponsoda, V., Scott, D., & Findlay, J. M. (1995). A probability vector and transition matrix analysis of eye movements during visual search. Acta Psychologica, 88(2), 167–185.
- Posner, M. I., & Cohen, Y. (1984). Components of visual orienting. Attention and Performance X, 32, 531-556.
- Pratt, J., Spalek, T. M., & Bradshaw, F. (1999). The time to detect targets at inhibited and noninhibited locations: Preliminary evidence for attentional momentum. *Journal of Experimental Psychology: Human Perception and Performance*, 25(3), 730–746.
- R.-Tavakoli, H., Atyabi, A., Rantanen, A., Laukka, S. J., Nefti-Meziani, S., & Heikkilä, J. (2015). Predicting the valence of a scene from observers' eye movements. *PLoS One*, 10(9), e0138198.
- Rafal, R. D., Machado, L., Ro, T., & Ingle, H. (2000). Looking forward to looking: Saccade preparation and the control of midbrain visuomotor reflexes. In S. Monsell & J. Driver (Eds.), *Control of cognitive performance: Attention and performance XVIII* (pp. 155–174). MIT Press.
- Rangelov, D., & Mattingley, J. B. (2020). Evidence accumulation during perceptual decision-making is sensitive to the dynamics of attentional selection. *Neuroimage*, 220, 117093.
- Ratcliff, R. (1978). A theory of memory retrieval. Psychological Review, 85(2), 59-108.
- Ratcliff, R., Van Zandt, T., & McKoon, G. (1999). Connectionist and diffusion models of reaction time. Psychological Review, 106(2), 261–300.
- Rayner, K. (1998). Eye movements in reading and information processing: 20 years of research. Psychological Bulletin, 124(3), 372–422.
- Rayner, K. (2009). Eye movements and attention in reading, scene perception, and visual search. The Quarterly Journal of Experimental Psychology, 62(8), 1457–1506.
- Rayner, K., Schotter, E. R., Masson, M. E. J., Potter, M. C., & Treiman, R. (2016). So much to read, so little time: How do we read, and can speed reading help? *Psychological Science in the Public Interest*, 17(1), 4–34.
- Rayner, K., Smith, T. J., Malcolm, G. L., & Henderson, J. M. (2009). Eye movements and visual encoding during scene perception. *Psychological Science*, 20(1), 6–10.
- Reinagel, P., & Zador, A. (1999). Natural scene statistics at the centre of gaze. Network, 10(4), 341-350.
- Rescorla, R. A., & Wagner, A. R. (1972). A theory of Pavlovian conditioning: Variations in the effectiveness of reinforcement and nonreinforcement. In A. H. Black & W. F. Prokasy (Eds.), *Classical Conditioning II: Current Research and Theory* (pp. 64–99). Appleton Century Crofts.
- Reutskaja, E., Nagel, R., Camerer, C. F., & Rangel, A. (2011). Search dynamics in consumer choice under time pressure: An eye-tracking study. *American Economic Review*, 101(2), 900–926.
- Roberts, J. H., & Lattin, J. M. (1997). Consideration: Review of research and prospects for future insights. *Journal of Marketing Research*, 34(3), 406–410.
- Rosch, E. (1978). Principles of categorization. In E. Rosch & B. B. Lloyd (Eds.), Cognition and categorization (pp. 27–48). Publishers: Lawrence Erlbaum Associates.
- Rousselet, G., Joubert, O., & Fabre-Thorpe, M. (2005). How long to get to the "gist" of real-world natural scenes? Visual Cognition, 12(6), 852–877.
- Ruan, S., MacEachern, S. N., Otter, T., & Dean, A. M. (2008). The dependent poisson race model and modeling dependence in conjoint choice experiments. *Psychometrika*, 73(2), 261–288.
- Russo, J. E., & Leclerc, F. (1994). An eye-fixation analysis of choice processes for consumer nondurables. Journal of Consumer Research, 21(2), 274–290.
- Russo, J. E., & Rosen, L. D. (1975). An eye fixation analysis of multialternative choice. *Memory and Cognition*, 3(3), 267–276.
- Rutishauser, U., & Koch, C. (2007). Probabilistic modeling of eye movement data during conjunction search via featurebased attention. *Journal of Vision*, 7(6), 1–20.

- Salvucci, D. D., & Goldberg, J. H. (2000). Identifying fixations and saccades in eye-tracking protocols. Proceedings of the Eye Tracking Research and Applications (ETRA) Symposium, (pp. 71–78).
- Sanders, A. F., & Donk, M. (1996). Visual Search. In O. Neumann & A. F. Sanders (Eds.), Handbook of perception and action (Vol. 3, pp. 43–77). Academic Press.
- Sapir, A., Soroker, N., Berger, A., & Henik, A. (1999). Inhibition of return in spatial attention: Direct evidence for collicular generation. *Nature Neuroscience*, 2(12), 1053–1054.
- Satomura, T., Wedel, M., & Pieters, R. (2014). Copy alert: A method and metric to detect visual copycat brands. *Journal of Marketing Research*, 51(1), 1–13.
- Sawaki, R., & Luck, S. J. (2010). Capture versus suppression of attention by salient singletons: Electrophysiological evidence for an automatic attend-to-me signal. Attention, Perception, & Psychophysics, 72(6), 1455–1470.
- Schotter, E. R., Berry, R. W., McKenzie, C. R. M., & Rayner, K. (2010). Gaze bias: Selective encoding and liking effects. Visual Cognition, 18(8), 1113–1132.
- Shen, J., & Palmeri, T. J. (2016). Modelling individual difference in visual categorization. Visual Cognition, 24(3), 260– 283.

Shi, S. W., & Trusov, M. (2021). The path to click: Are you on it? Marketing Science, 40(2), 344–365.

- Shi, S. W., Wedel, M., & Pieters, R. (2013). Information acquisition during online decision making: A model-based exploration using eye-tracking data. *Management Science*, 59(5), 1009–1026.
- Shimojo, S., Simion, C., Shimojo, E., & Scheier, C. (2003). Gaze bias both reflects and influences preference. *Nature Neuroscience*, 6(12), 1317–1322.
- Simola, J., Salojärvi, J., & Kojo, I. (2008). Using hidden Markov model to uncover processing states from eye movements in information search tasks. *Cognitive Systems Research*, 9(4), 237–251.
- Simon, H. A. (1955). A behavioral model of rational choice. The Quarterly Journal of Economics, 69(1), 99–118.
- Sims, C. A. (2003). Implications of rational inattention. Journal of Monetary Economics, 50(3), 665–690.
- Smith, P. L., & Ratcliff, R. (2004). Psychology and neurobiology of simple decisions. TRENDS in Neurosciences, 27(3), 161–168.
- Smith, T. J., & Henderson, J. M. (2009). Facilitation of return during scene viewing. Visual Cognition, 17(6-7), 1083-1108.
- Song, M., Wang, X., Zhang, H., & Li, J. (2019). Proactive information sampling in value-based decision-making: Deciding when and where to saccade. *Frontiers in human neuroscience*, 13, 35–35.
- Spalek, T. M., & Hammad, S. (2005). The left-to-right bias in inhibition of return is due to the direction of reading. *Psychological Science*, 16(1), 15–18.
- Stewart, N., Hermens, F., & Matthews, W. J. (2016). Eye movements in risky choice. Journal of Behavioral Decision Making, 29(2–3), 116–136.
- Stojić, H., Orquin, J. L., Dayan, P., Dolan, R. J., & Speekenbrink, M. (2020). Uncertainty in learning, choice, and visual fixation. Proceedings of the National Academy of Sciences, 117(6), 3291–3300.
- Stüttgen, P., Boatwright, P., & Monroe, R. T. (2012). A satisficing choice model. *Marketing Science*, 31(6), 878–899.
- Summerfield, C., & Blangero, A. (2017). Perceptual decision-making: What do we know, and what do we not know? In J.-C. Dreher & L. Tremblay (Eds.), *Decision neuroscience: An integrative perspective* (pp. 149–162). Elsevier Academic Press.
- Summerfield, C., & Tsetsos, K. (2012). Building bridges between perceptual and economic decision-making: Neural and computational mechanisms. *Frontiers in Neuroscience*, 6(70), 1–20.
- Swait, J., & Adamowicz, W. (2001). The influence of task complexity on consumer choice: A latent class model of decision strategy switching. *Journal of Consumer Research*, 28(1), 135–148.
- Tatler, B. W. (2007). The central fixation bias in scene viewing: Selecting an optimal viewing position independently of motor biases and image feature distributions. *Journal of Vision*, 7(14), 1–17.
- Tatler, B. W., & Vincent, B. T. (2009). The prominence of behavioural biases in eye guidance. *Visual Cognition*, 17(6–7), 1029–1054.
- Tavares, G., Perona, P., & Rangel, A. (2017). The attentional drift diffusion model of simple perceptual decision-making. Frontiers in Neuroscience, 11(468), 1–16.
- Thompson, K. G. (2005). Dissociation of selection from saccade programming. In L. Itti, G. Rees, & J. K. Tsotsos (Eds.), *Neurobiology of attention* (pp. 124–129). Elsevier.
- Tipper, S. P., Weaver, B., & Houghton, G. (1994). Behavioural goals determine inhibitory mechanisms of selective attention. *The Quarterly Journal of Experimental Psychology A: Human Experimental Psychology*, 47A(4), 809–840.
- Torralba, A., Oliva, A., Castelhano, M. S., & Henderson, J. M. (2006). Contextual guidance of eye movements and attention in real-world scenes: The role of global features in object search. *Psychological Review*, 113(4), 766–786.
- Towal, R. B., Mormann, M., & Koch, C. (2013). Simultaneous modeling of visual saliency and value computation improves predictions of economic choice. *Proceedings of the National Academy of Sciences of the United States*, 110(40), E3858–E3867.
- Townsend, C., & Kahn, B. E. (2013). The "Visual Preference Heuristic": The influence of visual versus verbal depiction on assortment processing, perceived variety, and choice overload. *Journal of Consumer Research*, 40 (5), 993–1015.
- Treisman, A., & Gelade, G. (1980). A feature-integration theory of attention. *Cognitive Psychology*, 12(1), 97–136.
 Treue, S. (2003). Visual attention: The where, what, how and why of saliency. *Current Opinion in Neurobiology*, 13(4), 428–432.
- Tversky, A. (1972). Elimination by Aspects: A theory of choice. Psychological Review, 79(4), 281-299.
- Ungerleider, L. G., & Mishkin, M. (1982). Two cortical visual systems. In D. J. Ingle, M. A. Goodale, & R. J. W. Mansfield (Eds.), Analysis of visual behavior (pp. 549–586). The MIT Press.

- Ursu, R. M., Zhang, Q., & Erdem, T. (2021). Prior information and consumer search: Evidence from eye-tracking. *Working Paper, NYU Stern School of Business*.
- Usher, M., & McClelland, J. L. (2001). The time course of perceptual choice: The leaky, competing accumulator model. *Psychological Review*, 108(3), 550–592.
- Usher, M., Olami, Z., & McClelland, J. L. (2002). Hick's law in a stochastic race model with speed-accuracy tradeoff. Journal of Mathematical Psychology, 46(6), 704–715.
- van der Lans, R., Pieters, R., & Wedel, M. (2008). Competitive brand salience. Marketing Science, 27(5), 452-461.
- van der Lans, R., Pieters, R., & Wedel, M. (2008). Eye-movement analysis of search effectiveness. Journal of the American Statistical Association, 103(482), 452–461.
- van der Lans, R., Pieters, R., & Wedel, M. (2021). Online advertising suppresses visual competition during planned purchases. *Journal of Consumer Research*, 48(3), 374–393.
- van der Lans, R., & Wedel, M. (2017). Eye movements during search and choice. In B. Wierenga & R. van der Lans (Eds.), *Handbook of marketing decision models* (2nd ed., pp. 331–359). Springer.
- van der Lans, R., Wedel, M., & Pieters, R. (2011). Defining eye-fixation sequences across individuals and tasks: The Binocular-Individual Threshold (BIT) Algorithm. *Behavior Research Methods*, 43(1), 239–257.
- van Herpen, E., & van Trijp, H. C. M. (2011). Front-of-pack nutrition labels. Their effect on attention and choices when consumers have varying goals and time constraints. *Appetite*, 57(1), 148–160.
- Van Loo, E. J., Grebitus, C., Nayga, R. M., Jr., Verbeke, W., & Roosen, J. (2018). On the measurement of consumer preferences and food choice behavior: The relation between visual attention and choices. *Applied Economic Perspectives* and Policy, 40(4), 538–562.
- Venkatraman, V., Payne, J. W., & Huettel, S. A. (2014). An overall probability of winning heuristic for complex risky decisions: Choice and eye fixation evidence. Organizational Behavior and Human Decision Processes, 125(2), 73–87.
- Von Neumann, J., & Morgenstern, O. (1947). Theory of games and economic behavior, 2nd rev. ed. Princeton University Press.
- Voss, A., Nagler, M., & Lerche, V. (2013). Diffusion models in experimental psychology: A practical introduction. *Experimental Psychology*, 60(6), 385–402.
- Voss, A., Rothermund, K., & Voss, J. (2004). Interpreting the parameters of the diffusion model: An empirical validation. *Memory & Cognition*, 32(7), 1206–1220.
- Wang, X.-J. (2002). Probabilistic decision making by slow reverberation in cortical circuits. *Neuron*, 36(5), 955–968.
- Webb, R. (2019). The (Neural) dynamics of stochastic choice. Management Science, 65(1), 230–255.
- Wedel, M., & Pieters, R. (2000). Eye fixations on advertisements and memory for brands: A model and findings. *Marketing Science*, 19(4), 297–312.
- Wedel, M., & Pieters, R. (2008). Eye-tracking for visual marketing. Foundations and Trends(R) in Marketing, 1(4), 231– 320.
- Wedel, M., Pieters, R., & Liechty, J. (2008). Attention switching during scene perception: How goals influence the time course of eye movements across advertisements. *Journal of Experimental Psychology: Applied*, 14(2), 129–138.
- Wedel, M., Yan, J., Siegel, E. L., & Li, H. A. (2016). Nodule detection with eye movements. Journal of Behavioral Decision Making, 29(2–3), 254–270.
- Weitzman, M. L. (1979). Optimal search for the best alternative. Econometrica, 47(3), 641-654.
- Wolfe, J. M. (1994). Guided Search 2.0: A revised model of visual search. Psychonomic Bulletin & Review 1(2), 202-238.
- Wolfe, J. M. (1998). Visual search. In H. Pashler (Ed.), Attention (pp. 13-73). University College London Press Ltd.
- Yang, L. C., Toubia, O., & de Jong, M. G. (2015). A bounded rationality model of information search and choice in preference measurement. *Journal of Marketing Research*, 52(2), 166–183.
- Yang, L. C., Toubia, O., & de Jong, M. G. (2018). Attention, information processing, and choice in incentive-aligned choice experiments. *Journal of Marketing Research*, 55(6), 783–800.
- Yarbus, A. L. (1967). Eye movements during perception of complex objects (B. Haigh, Trans.). In Eye movements and vision (pp. 171-211). Plenum Press.
- Yegoryan, N., Guhl, D., & Klapper, D. (2020). Inferring attribute non-attendance using eye tracking in choice-based conjoint analysis. *Journal of Business Research*, 111, 290–304.
- Zelinsky, G. J. (2008). A theory of eye movements during target acquisition. Psychological Review, 115(4), 787-835.
- Zelinsky, G. J., & Sheinberg, D. L. (1997). Eye movements during parellel-serial visual search. Journal of Experimental Psychology: Human Perception and Performance, 23(1), 244–262.
- Zhang, J., Wedel, M., & Pieters, R. (2009). Sales effects of attention to feature advertisements: A Bayesian mediation analysis. *Journal of Marketing Research*, 46(5), 669–681.
- Zhang, Q. P., Ursu, R. M., & Erdem, T. (2020). Prior information and consumer search: Evidence from eye-tracking. Marketing Science Institute Working Paper Series, Report No. 20-141.

Manuscript Received: 21 SEP 2021 Final Version Received: 15 JUN 2022 Accepted: 16 JUN 2022 Published Online Date: 19 JUL 2022